IAEA Safety Standards for protecting people and the environment

Ageing Management for Research Reactors

Specific Safety Guide No. SSG-10





IAEA SAFETY RELATED PUBLICATIONS

IAEA SAFETY STANDARDS

Under the terms of Article III of its Statute, the IAEA is authorized to establish or adopt standards of safety for protection of health and minimization of danger to life and property, and to provide for the application of these standards.

The publications by means of which the IAEA establishes standards are issued in the **IAEA Safety Standards Series**. This series covers nuclear safety, radiation safety, transport safety and waste safety. The publication categories in the series are **Safety Fundamentals**, **Safety Requirements** and **Safety Guides**.

Information on the IAEA's safety standards programme is available at the IAEA Internet site

http://www-ns.iaea.org/standards/

The site provides the texts in English of published and draft safety standards. The texts of safety standards issued in Arabic, Chinese, French, Russian and Spanish, the IAEA Safety Glossary and a status report for safety standards under development are also available. For further information, please contact the IAEA at PO Box 100, 1400 Vienna, Austria.

All users of IAEA safety standards are invited to inform the IAEA of experience in their use (e.g. as a basis for national regulations, for safety reviews and for training courses) for the purpose of ensuring that they continue to meet users' needs. Information may be provided via the IAEA Internet site or by post, as above, or by email to Official.Mail@iaea.org.

OTHER SAFETY RELATED PUBLICATIONS

The IAEA provides for the application of the standards and, under the terms of Articles III and VIII.C of its Statute, makes available and fosters the exchange of information relating to peaceful nuclear activities and serves as an intermediary among its Member States for this purpose.

Reports on safety and protection in nuclear activities are issued as **Safety Reports**, which provide practical examples and detailed methods that can be used in support of the safety standards.

Other safety related IAEA publications are issued as **Radiological Assessment Reports**, the International Nuclear Safety Group's **INSAG Reports**, **Technical Reports** and **TECDOCs**. The IAEA also issues reports on radiological accidents, training manuals and practical manuals, and other special safety related publications. Security related publications are issued in the **IAEA Nuclear Security Series**.

AGEING MANAGEMENT FOR RESEARCH REACTORS

Safety standards survey

The IAEA welcomes your response. Please see: http://www-ns.iaea.org/standards/feedback.htm The following States are Members of the International Atomic Energy Agency:

AFGHANISTAN ALBANIA ALGERIA ANGOLA ARGENTINA ARMENIA AUSTRALIA AUSTRIA AZERBAIJAN BAHRAIN BANGLADESH BELARUS BELGIUM BELIZE BENIN BOLIVIA BOSNIA AND HERZEGOVINA BOTSWANA BRAZIL BULGARIA BURKINA FASO BURUNDI CAMBODIA CAMEROON CANADA CENTRAL AFRICAN REPUBLIC CHAD CHILE CHINA COLOMBIA CONGO COSTA RICA CÔTE D'IVOIRE CROATIA CUBA CYPRUS CZECH REPUBLIC DEMOCRATIC REPUBLIC OF THE CONGO DENMARK DOMINICAN REPUBLIC ECUADOR EGYPT EL SALVADOR ERITREA ESTONIA **ETHIOPIA** FINLAND FRANCE GABON GEORGIA GERMANY

GHANA GREECE GUATEMALA HAITI HOLY SEE HONDURAS HUNGARY ICELAND INDIA INDONESIA IRAN, ISLAMIC REPUBLIC OF IRAO IRELAND ISRAEL ITALY JAMAICA IAPAN JORDAN KAZAKHSTAN KENYA KOREA, REPUBLIC OF KUWAIT KYRGYZSTAN LATVIA LEBANON LESOTHO LIBERIA LIBYAN ARAB JAMAHIRIYA LIECHTENSTEIN LITHUANIA LUXEMBOURG MADAGASCAR MALAWI MALAYSIA MALI MALTA MARSHALL ISLANDS MAURITANIA MAURITIUS MEXICO MONACO MONGOLIA MONTENEGRO MOROCCO MOZAMBIQUE MYANMAR NAMIBIA NEPAL NETHERLANDS NEW ZEALAND NICARAGUA NIGER NIGERIA

NORWAY OMAN PAKISTAN PALAU PANAMA PARAGUAY PERU PHILIPPINES POLAND PORTUGAL QATAR REPUBLIC OF MOLDOVA ROMANIA RUSSIAN FEDERATION SAUDI ARABIA SENEGAL SERBIA SEYCHELLES SIERRA LEONE SINGAPORE SLOVAKIA **SLOVENIA** SOUTH AFRICA SPAIN SRI LANKA SUDAN SWEDEN SWITZERLAND SYRIAN ARAB REPUBLIC TAIIKISTAN THAILAND THE FORMER YUGOSLAV REPUBLIC OF MACEDONIA TUNISIA TURKEY UGANDA UKRAINE UNITED ARAB EMIRATES UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND UNITED REPUBLIC OF TANZANIA UNITED STATES OF AMERICA URUGUAY UZBEKISTAN VENEZUELA VIETNAM YEMEN ZAMBIA ZIMBABWE

The Agency's Statute was approved on 23 October 1956 by the Conference on the Statute of the IAEA held at United Nations Headquarters, New York; it entered into force on 29 July 1957. The Headquarters of the Agency are situated in Vienna. Its principal objective is "to accelerate and enlarge the contribution of atomic energy to peace, health and prosperity throughout the world".

IAEA SAFETY STANDARDS SERIES No. SSG-10

AGEING MANAGEMENT FOR RESEARCH REACTORS

SPECIFIC SAFETY GUIDE

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA, 2010

COPYRIGHT NOTICE

All IAEA scientific and technical publications are protected by the terms of the Universal Copyright Convention as adopted in 1952 (Berne) and as revised in 1972 (Paris). The copyright has since been extended by the World Intellectual Property Organization (Geneva) to include electronic and virtual intellectual property. Permission to use whole or parts of texts contained in IAEA publications in printed or electronic form must be obtained and is usually subject to royalty agreements. Proposals for non-commercial reproductions and translations are welcomed and considered on a case-by-case basis. Enquiries should be addressed to the IAEA Publishing Section at:

Marketing and Sales Unit, Publishing Section International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria fax: +43 1 2600 29302 tel.: +43 1 2600 22417 email: sales.publications@iaea.org http://www.iaea.org/books

> © IAEA, 2010 Printed by the IAEA in Austria October 2010 STI/PUB/1447

IAEA Library Cataloguing in Publication Data

Ageing management for research reactors : specific safety guide. — Vienna : International Atomic Energy Agency, 2010. p. ; 24 cm. — (IAEA safety standards series, ISSN 1020–525X ; no. SSG-10) STI/PUB/1447 ISBN 978–92–0–102810–5 Includes bibliographical references.

1. Research reactors. 2. Nuclear reactors — Decommissioning. 3. Nuclear reactors — Safety measures. 4. Radiation protection — Government policy. I. International Atomic Energy Agency. II. Series.

IAEAL

10-00638

FOREWORD

The IAEA's Statute authorizes the Agency to establish safety standards to protect health and minimize danger to life and property — standards which the IAEA must use in its own operations, and which a State can apply by means of its regulatory provisions for nuclear and radiation safety. A comprehensive body of safety standards under regular review, together with the IAEA's assistance in their application, has become a key element in a global safety regime.

In the mid-1990s, a major overhaul of the IAEA's safety standards programme was initiated, with a revised oversight committee structure and a systematic approach to updating the entire corpus of standards. The new standards that have resulted are of a high calibre and reflect best practices in Member States. With the assistance of the Commission on Safety Standards, the IAEA is working to promote the global acceptance and use of its safety standards.

Safety standards are only effective, however, if they are properly applied in practice. The IAEA's safety services — which range in scope from engineering safety, operational safety, and radiation, transport and waste safety to regulatory matters and safety culture in organizations — assist Member States in applying the standards and appraise their effectiveness. These safety services enable valuable insights to be shared and all Member States are urged to make use of them.

Regulating nuclear and radiation safety is a national responsibility, and many Member States have decided to adopt the IAEA's safety standards for use in their national regulations. For the contracting parties to the various international safety conventions, IAEA standards provide a consistent, reliable means of ensuring the effective fulfilment of obligations under the conventions. The standards are also applied by designers, manufacturers and operators around the world to enhance nuclear and radiation safety in power generation, medicine, industry, agriculture, research and education.

The IAEA takes seriously the enduring challenge for users and regulators everywhere: that of ensuring a high level of safety in the use of nuclear materials and radiation sources around the world. Their continuing utilization for the benefit of humankind must be managed in a safe manner, and the IAEA safety standards are designed to facilitate the achievement of that goal.

THE IAEA SAFETY STANDARDS

BACKGROUND

Radioactivity is a natural phenomenon and natural sources of radiation are features of the environment. Radiation and radioactive substances have many beneficial applications, ranging from power generation to uses in medicine, industry and agriculture. The radiation risks to workers and the public and to the environment that may arise from these applications have to be assessed and, if necessary, controlled.

Activities such as the medical uses of radiation, the operation of nuclear installations, the production, transport and use of radioactive material, and the management of radioactive waste must therefore be subject to standards of safety.

Regulating safety is a national responsibility. However, radiation risks may transcend national borders, and international cooperation serves to promote and enhance safety globally by exchanging experience and by improving capabilities to control hazards, to prevent accidents, to respond to emergencies and to mitigate any harmful consequences.

States have an obligation of diligence and duty of care, and are expected to fulfil their national and international undertakings and obligations.

International safety standards provide support for States in meeting their obligations under general principles of international law, such as those relating to environmental protection. International safety standards also promote and assure confidence in safety and facilitate international commerce and trade.

A global nuclear safety regime is in place and is being continuously improved. IAEA safety standards, which support the implementation of binding international instruments and national safety infrastructures, are a cornerstone of this global regime. The IAEA safety standards constitute a useful tool for contracting parties to assess their performance under these international conventions.

THE IAEA SAFETY STANDARDS

The status of the IAEA safety standards derives from the IAEA's Statute, which authorizes the IAEA to establish or adopt, in consultation and, where appropriate, in collaboration with the competent organs of the United Nations and with the specialized agencies concerned, standards of safety for protection of health and minimization of danger to life and property, and to provide for their application.

With a view to ensuring the protection of people and the environment from harmful effects of ionizing radiation, the IAEA safety standards establish fundamental safety principles, requirements and measures to control the radiation exposure of people and the release of radioactive material to the environment, to restrict the likelihood of events that might lead to a loss of control over a nuclear reactor core, nuclear chain reaction, radioactive source or any other source of radiation, and to mitigate the consequences of such events if they were to occur. The standards apply to facilities and activities that give rise to radiation risks, including nuclear installations, the use of radiation and radioactive sources, the transport of radioactive material and the management of radioactive waste.

Safety measures and security measures¹ have in common the aim of protecting human life and health and the environment. Safety measures and security measures must be designed and implemented in an integrated manner so that security measures do not compromise safety and safety measures do not compromise security.

The IAEA safety standards reflect an international consensus on what constitutes a high level of safety for protecting people and the environment from harmful effects of ionizing radiation. They are issued in the IAEA Safety Standards Series, which has three categories (see Fig. 1).

Safety Fundamentals

Safety Fundamentals present the fundamental safety objective and principles of protection and safety, and provide the basis for the safety requirements.

Safety Requirements

An integrated and consistent set of Safety Requirements establishes the requirements that must be met to ensure the protection of people and the environment, both now and in the future. The requirements are governed by the objective and principles of the Safety Fundamentals. If the requirements are not met, measures must be taken to reach or restore the required level of safety. The format and style of the requirements facilitate their use for the establishment, in a harmonized manner, of a national regulatory framework. Requirements, including numbered 'overarching' requirements, are expressed

¹ See also publications issued in the IAEA Nuclear Security Series.



FIG. 1. The long term structure of the IAEA Safety Standards Series.

as 'shall' statements. Many requirements are not addressed to a specific party, the implication being that the appropriate parties are responsible for fulfilling them.

Safety Guides

Safety Guides provide recommendations and guidance on how to comply with the safety requirements, indicating an international consensus that it is necessary to take the measures recommended (or equivalent alternative measures). The Safety Guides present international good practices, and increasingly they reflect best practices, to help users striving to achieve high levels of safety. The recommendations provided in Safety Guides are expressed as 'should' statements.

APPLICATION OF THE IAEA SAFETY STANDARDS

The principal users of safety standards in IAEA Member States are regulatory bodies and other relevant national authorities. The IAEA safety

standards are also used by co-sponsoring organizations and by many organizations that design, construct and operate nuclear facilities, as well as organizations involved in the use of radiation and radioactive sources.

The IAEA safety standards are applicable, as relevant, throughout the entire lifetime of all facilities and activities — existing and new — utilized for peaceful purposes and to protective actions to reduce existing radiation risks. They can be used by States as a reference for their national regulations in respect of facilities and activities.

The IAEA's Statute makes the safety standards binding on the IAEA in relation to its own operations and also on States in relation to IAEA assisted operations.

The IAEA safety standards also form the basis for the IAEA's safety review services, and they are used by the IAEA in support of competence building, including the development of educational curricula and training courses.

International conventions contain requirements similar to those in the IAEA safety standards and make them binding on contracting parties. The IAEA safety standards, supplemented by international conventions, industry standards and detailed national requirements, establish a consistent basis for protecting people and the environment. There will also be some special aspects of safety that need to be assessed at the national level. For example, many of the IAEA safety standards, in particular those addressing aspects of safety in planning or design, are intended to apply primarily to new facilities and activities. The requirements established in the IAEA safety standards might not be fully met at some existing facilities that were built to earlier standards. The way in which IAEA safety standards are to be applied to such facilities is a decision for individual States.

The scientific considerations underlying the IAEA safety standards provide an objective basis for decisions concerning safety; however, decision makers must also make informed judgements and must determine how best to balance the benefits of an action or an activity against the associated radiation risks and any other detrimental impacts to which it gives rise.

DEVELOPMENT PROCESS FOR THE IAEA SAFETY STANDARDS

The preparation and review of the safety standards involves the IAEA Secretariat and four safety standards committees, for nuclear safety (NUSSC), radiation safety (RASSC), the safety of radioactive waste (WASSC) and the safe transport of radioactive material (TRANSSC), and a Commission on Safety Standards (CSS) which oversees the IAEA safety standards programme (see Fig. 2).



FIG. 2. The process for developing a new safety standard or revising an existing standard.

All IAEA Member States may nominate experts for the safety standards committees and may provide comments on draft standards. The membership of the Commission on Safety Standards is appointed by the Director General and includes senior governmental officials having responsibility for establishing national standards.

A management system has been established for the processes of planning, developing, reviewing, revising and establishing the IAEA safety standards. It articulates the mandate of the IAEA, the vision for the future application of the safety standards, policies and strategies, and corresponding functions and responsibilities.

INTERACTION WITH OTHER INTERNATIONAL ORGANIZATIONS

The findings of the United Nations Scientific Committee on the Effects of Atomic Radiation (UNSCEAR) and the recommendations of international

expert bodies, notably the International Commission on Radiological Protection (ICRP), are taken into account in developing the IAEA safety standards. Some safety standards are developed in cooperation with other bodies in the United Nations system or other specialized agencies, including the Food and Agriculture Organization of the United Nations, the United Nations Environment Programme, the International Labour Organization, the OECD Nuclear Energy Agency, the Pan American Health Organization and the World Health Organization.

INTERPRETATION OF THE TEXT

Safety related terms are to be understood as defined in the IAEA Safety Glossary (see http://www-ns.iaea.org/standards/safety-glossary.htm). Otherwise, words are used with the spellings and meanings assigned to them in the latest edition of The Concise Oxford Dictionary. For Safety Guides, the English version of the text is the authoritative version.

The background and context of each standard in the IAEA Safety Standards Series and its objective, scope and structure are explained in Section 1, Introduction, of each publication.

Material for which there is no appropriate place in the body text (e.g. material that is subsidiary to or separate from the body text, is included in support of statements in the body text, or describes methods of calculation, procedures or limits and conditions) may be presented in appendices or annexes.

An appendix, if included, is considered to form an integral part of the safety standard. Material in an appendix has the same status as the body text, and the IAEA assumes authorship of it. Annexes and footnotes to the main text, if included, are used to provide practical examples or additional information or explanation. Annexes and footnotes are not integral parts of the main text. Annex material published by the IAEA is not necessarily issued under its authorship; material under other authorship may be presented in annexes to the safety standards. Extraneous material presented in annexes is excerpted and adapted as necessary to be generally useful.

CONTENTS

1.	INTRODUCTION	1
	Background (1.1–1.4).	1 2
	Scope (17–112)	2
	Structure (1.13)	4
2.	MANAGEMENT SYSTEM FOR AGEING MANAGEMENT	4
	General (2.1–2.4)	4
	Management responsibility (2.5–2.8)	6
	Resource management (2.9–2.10)	7
	Process implementation (2.11–2.16)	7
	Measurement, assessment and improvement (2.17–2.19)	8
3.	AGEING AND SAFETY OF RESEARCH REACTORS	9
	Basic concepts (3.1–3.10)	9
	Ageing and defence in depth (3.11–3.13)	11
	Service conditions and ageing (3.14–3.16)	12
4.	AGEING CONSIDERATIONS IN DIFFERENT STAGES	
	OF THE LIFETIME OF A RESEARCH REACTOR	14
	General (4.1–4.4)	14
	Design (4.5–4.11).	15
	Fabrication and construction (4.12)	17
	Commissioning (4.13–4.14).	17
	Operation (4.15–4.19)	18
	Utilization and modifications (4.20–4.23)	20
	Extended shutdown (4.24–4.26)	20
	Decommissioning (4.27)	21
5.	ELEMENTS OF AGEING MANAGEMENT PROGRAMMES	22
	General (5.1–5.3)	22
	Screening of SSCs for ageing management review (5.4–5.8)	22
	Identification and understanding of ageing degradation (5.9–5.11)	24

	Minim	ization of expected ageing degradation (5.12–5.13)	25
	Detect	10n, monitoring and trending of ageing degradation	25
	Mitiga	tion of ageing degradation $(5.23-5.25)$	27
	Contin	hous improvement of the ageing management programme	21
	(5.2	26–5 30)	28
	Record	$1 \text{ keening } (5 \ 31 - 5 \ 33)$	20
	Record	r kcoping (5.51 - 5.55)	2)
6.	MANA	AGEMENT OF OBSOLESCENCE (6.1–6.8)	30
7.	INTEF	RFACES WITH OTHER TECHNICAL AREAS	32
	τ	(7.1)	22
	Introdu	(7.1)	32
	Daniad	enance, periodic testing and inspection $(7.2-7.3)$	32
	Feriod	Ic safety review $(/.4-/./)$	33 24
	Equipr		34
	Config	stitution of the design basis for SSCs $(7.9-7.10)$	34
	Contin	viol as for amountainer (7.12, 7.12)	33 25
	Contin	$\frac{1}{2} = \frac{1}{2} = \frac{1}$	30
	Post-se	ervice surveillance and testing (7.14)	36
REF	ERENC	CES	37
ANN	JEX I	EFFECT OF AGEING FOR DIFFERENT SERVICE	
		CONDITIONS	39
ANN	JEX II:	EXAMPLE FOR SCREENING OF RESEARCH REACTOR	
		SSCs FOR AGEING MANAGEMENT PURPOSES	42
CON	TRIBU	TORS TO DRAFTING AND REVIEW	47
BOD	DIES FC	OR THE ENDORSEMENT OF	
IA	AEA SA	FETY STANDARDS	49

1. INTRODUCTION

BACKGROUND

1.1. This Safety Guide was developed under the IAEA programme for safety standards for research reactors, which covers all the important areas of research reactor safety. It supplements and elaborates upon the safety requirements for ageing management of research reactors that are established in paras 6.68–6.70 and 7.109 of the IAEA Safety Requirements publication, Safety of Research Reactors [1].

1.2. The safety of a research reactor requires that provisions be made in its design to facilitate ageing management. Throughout the lifetime of a research reactor, including its decommissioning, ageing management of its structures, systems and components (SSCs) important to safety¹ is required, to ensure continued adequacy of the safety level, reliable operation of the reactor, and compliance with the operational limits and conditions.

1.3. Managing the safety aspects of research reactor ageing requires implementation of an effective programme for the monitoring, prediction, and timely detection and mitigation of degradation of SSCs important to safety, and for maintaining their integrity and functional capability throughout their service lives.

1.4. Ageing management is defined as engineering, operation, and maintenance strategy and actions to control within acceptable limits the ageing degradation of SSCs [2]. Ageing management includes activities such as repair, refurbishment and replacement of SSCs, which are similar to other activities carried out at a research reactor in maintenance and testing or when a modification project takes place. However, it is important to recognize that effective management of ageing

¹ Items important to safety: An item that is part of a safety group and/or whose malfunction or failure could lead to radiation exposure of the site personnel or members of the public. Items important to safety include:

⁽a) Those SSCs whose malfunction or failure could lead to undue radiation exposure to site personnel or members of the public;

⁽b) Those SSCs that prevent anticipated operational occurrences from leading to accident conditions;

⁽c) Those features that are provided to mitigate the consequences of malfunction or failure of SSCs [2].

requires the use of a methodology that will detect and evaluate ageing degradation as a consequence of the service conditions², and involves the application of countermeasures for prevention and mitigation of ageing degradation.

OBJECTIVE

1.5. The objective of this Safety Guide is to provide recommendations on managing ageing of SSCs important to safety at research reactors on the basis of international good practice. This Safety Guide is intended for use by operating organizations in establishing, implementing and improving ageing management programmes for research reactors. It may be used by regulatory bodies in verifying that ageing of research reactors is being effectively managed.

1.6. While the intention is not to recommend ways in which to make the operation of a research reactor more effective or more efficient, the recommendations provided in this Safety Guide on managing ageing of SSCs to ensure safety will undoubtedly contribute to the enhancement of their long term reliability and availability.

SCOPE

1.7. This Safety Guide covers all aspects relating to the establishment, implementation and continuous improvement of an ageing management programme. It not only focuses on managing the physical ageing of SSCs important to safety but also provides guidance on safety aspects of managing obsolescence (non-physical ageing).

1.8. In formulating the recommendations in this Safety Guide, consideration has been given to the recommendations in the IAEA Safety Guide, Ageing Management for Nuclear Power Plants [3]. Where appropriate, in consideration of differences in hazard potential and complexity of systems between nuclear power plants and research reactors, certain provisions of Ref. [3] have been adopted.

² Service conditions are actual physical states or influences during the service lives of SSCs, including operating conditions (normal and erroneous included), design basis event conditions, and conditions following a design basis event [2].

1.9. The recommendations provided in this Safety Guide are intended to be applicable to research reactors having limited hazard potential to the public and the environment. It deals with all types of heterogeneous, thermal spectrum, research reactors having a power rating of up to several tens of megawatts. Additional guidance may be necessary for research reactors of higher power, specialized reactors (e.g. fast spectrum reactors and homogeneous reactors) and reactors with specialized facilities (e.g. hot or cold neutron sources, or high pressure and high temperature test loops).

1.10. Research reactors having a power rating of up to several tens of kilowatts and critical assemblies may need a less comprehensive ageing management programme than that outlined here. While consideration should be given to all the recommendations made in this Safety Guide, some may not be applicable to these low power research reactors. For these reasons, the recommendations in this Safety Guide should be graded³ for their applicability to the specific research reactor under consideration (see Ref. [1], paras 1.11-1.14).

1.11. A graded approach may be applied in determining the appropriate frequency of inspections, in selecting detection methods, as well as in establishing measures for prevention and mitigation of ageing effects. The graded approach may be based on the estimated service lives of the SSCs, their complexity and their ease of replacement. A graded approach may also be applicable to the resources necessary to implement the ageing management programme. While a dedicated organizational unit may be needed to implement the ageing management programme for higher power research reactors, the ageing management activities for research reactors having a low power may be performed by the maintenance personnel. A Safety Guide on the use of a graded approach in the application of the safety requirements for research reactors is in preparation.

1.12. Although managing succession for the personnel of research reactors and knowledge management are important issues in relation to safety at research reactors, they are outside the scope of this Safety Guide. Recommendations relating to the ageing of research reactor personnel are provided in Ref. [4].

³ The recommendations should be graded, for example, by considering — using sound engineering judgement — the operational importance of the item or activity, its importance to safety, and its maturity and complexity.

STRUCTURE

1.13. This Safety Guide consists of seven sections and two annexes. Section 2 covers the management system for ageing management. Section 3 presents the basic concepts and explanations of the terms used in this Safety Guide, and an overview of ageing effects and their relation to safety at research reactors. Section 4 provides recommendations on proactive considerations on ageing during the various stages of the lifetime of a research reactor. Section 5 sets out the elements of a systematic ageing management programme, and provides advice on implementation of the associated actions or activities. Section 6 presents recommendations on management of SSCs during their obsolescence. Section 7 discusses different technical aspects relating to ageing management, such as maintenance, periodic testing and inspection, periodic safety review, equipment qualification, design basis reconstitution, configuration management and continued safe operation. Annex I presents ageing mechanisms and their effects for different categories of service conditions. Annex II provides an example of screening of SSCs for ageing management for open pool, light water and heavy water moderator research reactors.

2. MANAGEMENT SYSTEM FOR AGEING MANAGEMENT⁴

GENERAL

2.1. A documented management system that integrates the safety, health, environmental, security, quality and economic objectives of the operating organization of a research reactor is required to be in place [5]. The documentation of the management system should describe the system that controls the planning and implementation of all activities at the research reactor throughout its lifetime, including ageing management activities. Approval of the management system (or parts thereof) by the regulatory body may be required. The management system should cover four functional categories: management responsibility; resource

⁴ The recommendations provided in this section can also be used, as applicable, to establish a management system for managing the obsolescence of SSCs.

management; process implementation; and measurement, assessment and improvement. In general:

- (1) Management responsibility includes the support and commitment of management necessary to achieve the objectives of the operating organization.
- (2) Resource management includes the measures necessary to ensure that the resources essential to the implementation of strategy and the achievement of the objectives of the operating organization are identified and made available.
- (3) Process implementation includes the activities and tasks necessary to achieve the goals of the organization.
- (4) Measurement and assessment provide an indication of the effectiveness of management processes and work performance compared with objectives or benchmarks; it is through measurement and assessment that opportunities for improvement are identified.

The requirements for the management system are established in Ref. [1], paras 4.5–4.13, and in Ref. [5], and further recommendations are provided in Ref. [6].

2.2. As part of the integrated management system, a system for ageing management should be established and put into effect by the operating organization. For a new research reactor project, the management system for ageing management should be established early on in the design stage. The system should cover all items, services and processes important to safety and should include a means of establishing control over ageing management activities, thereby providing confidence that they are performed according to the established requirements. In determining how the requirements of the management system for ageing management are to be applied, a graded approach based on the relative importance to safety of each item or process should be used.

2.3. The objective of the management system as applied to ageing management is to ensure that the facility meets the requirements for safety as derived from:

- The requirements of the regulatory body;
- Design requirements and assumptions;
- The safety analysis report;
- Operational limits and conditions;
- The administrative requirements established by the management of the research reactor.

2.4. The management system should support the development, implementation and enhancement of a strong safety culture in all aspects of the ageing management programme.

MANAGEMENT RESPONSIBILITY

2.5. The management system for ageing management should provide a framework for managing, performing and assessing the activities necessary to prevent, detect, monitor, assess and mitigate ageing effects. The documentation of the management system for ageing management should include descriptions of the organizational structure, functional responsibilities, levels of authority and interactions of those managing, performing and assessing the adequacy of the ageing management activities. It should also cover other management measures, including planning, scheduling, resource allocation and human factors.

2.6. The management system for ageing management should be outlined in a description of the ageing management programme and should be documented in operating procedures. The operating procedures should address all applicable requirements specified in the integrated management system established by the operating organization.

2.7. The ageing management activities should be planned and performed, and their results recorded, in accordance with approved procedures and instructions, which should be included in the documentation of the management system for ageing management.

2.8. Successful implementation of the ageing management programme requires:

- (a) Planning and prioritization of work;
- (b) Addressing all relevant regulatory requirements, codes and standards;
- (c) Addressing the requirements derived from the operational limits and conditions, and from the safety analysis report;
- (d) Ensuring the availability of spare parts, special tools and equipment;
- (e) Following up inspection and test results in a timely fashion;
- (f) Ensuring the provision of qualified personnel with suitable skills;
- (g) Establishing appropriate operating procedures following relevant standards, including procedures for assessing and correcting non-conforming items;
- (h) Identifying, disseminating and using information on good practices from designers, manufacturers and other operating organizations;
- (i) Performing and adequately documenting the required inspections and tests;

(j) Performing root cause analyses of significant degradation of SSCs and incorporating lessons learned from experience.

RESOURCE MANAGEMENT

2.9. The operating organization should provide adequate resources (both human resources and financial resources) to execute the ageing management programme. The management of the operating organization, in particular the reactor manager, should participate in the ageing management activities by:

- (a) Determining the required staff competences and providing training where appropriate;
- (b) Preparing and issuing specifications and procedures for the ageing management programme;
- (c) Having frequent personal contact with staff, including observation of work in progress;
- (d) Supervising external personnel (including suppliers) who perform ageing management activities, and ensuring that these personnel are adequately trained and qualified;
- (e) Supporting and participating in ageing evaluations.

2.10. The management system for ageing management should include provisions to ensure that the ageing management activities are planned, performed and controlled in a manner that ensures effective communication and clear assignment of responsibility.

PROCESS IMPLEMENTATION

2.11. The operating organization should nominate the person who has the responsibility and accountability for implementing the ageing management programme. This person is usually the reactor manager or a staff member acting on behalf of the reactor manager.

2.12. Significant changes or failures of any SSCs important to safety should be recorded, and the data should be analysed and trends discerned, to identify the causes of these changes or failures. Where the causes of failures have been determined, the information should be used as input to improvement of the ageing management programme.

2.13. The scope and frequency of tests and in-service inspections should be specified and they should be consistent with the operational limits and conditions and regulatory requirements. The recording and presentation of test results should permit easy comparison with the results of previous inspections and tests, to enable detection of any changes since previous tests and any deviations from reference values (baseline data).

2.14. Valid monitoring and measurements should be performed to provide evidence of conformity to requirements and satisfactory performance in service.

2.15. Equipment used for monitoring, data collection, and inspections and tests should be calibrated. The equipment calibration should be documented to demonstrate that this calibration is up to date. Inspection and tests should be performed by qualified personnel and be in accordance with approved procedures.

2.16. The management system for ageing management should include measures to control records essential to the performance of ageing management activities and to verification of the results achieved. The records process should provide for the identification, approval, review, filing, retrieval and disposal of records.

MEASUREMENT, ASSESSMENT AND IMPROVEMENT

2.17. Measures should be established to ensure that ageing management activities are accomplished as specified in the appropriate procedures. Such measures should include:

- Reviews of procedures;
- Verification by inspection, witnessing and surveillance;
- Checks of non-conformances and implementation of corrective actions;
- Follow-up of the adequacy and timeliness of corrective actions.

2.18. An independent assessment of the ageing management programme should be conducted. This task may be performed by the safety committee (see paras 4.15, 7.25 and 7.26 of Ref. [1]) or by another competent body.

2.19. Audits should be performed to determine the adequacy and effectiveness of all aspects of the implementation of the ageing management programme and its adherence to the requirements of the management system for ageing management. The operating organization should evaluate the results of

independent assessment, including audits, and should take any necessary actions to make improvements.

3. AGEING AND SAFETY OF RESEARCH REACTORS

BASIC CONCEPTS

3.1. Ageing is defined as a general process in which the characteristics of SSCs gradually change with time or use [2]. Research reactors experience two kinds of time dependent changes:

- (1) Degradation of SSCs (physical ageing), i.e. gradual deterioration in their physical characteristics;
- (2) Obsolescence of SSCs (non-physical ageing), i.e. their becoming out of date in comparison with current knowledge, standards and technology.

3.2. Physical ageing eventually leads to degradation of materials subjected to normal service conditions. These include the normal operational conditions under which SSCs are required to operate and the anticipated operational occurrences under which SSCs are required to continue functioning properly. The ageing effects of accident conditions (design basis accidents and beyond design basis accidents) need to be evaluated on a case-by-case basis.

3.3. Degradation may result in a reduction or loss of the ability of SSCs to function within their acceptance criteria. The safety and utilization of the facility will be affected unless preventive measures are in place and timely appropriate corrective actions are taken.

3.4. Physical ageing reduces the safety margins provided in the design of SSCs. If reductions in these margins are not detected, research reactor safety could be compromised unless mitigatory actions are taken before loss of functional capability occurs.

- 3.5. During the lifetime of a research reactor:
- (a) Advances will occur in technology, resulting in the introduction of new components or techniques. This may lead to difficulties in obtaining spare

parts. Introduction of new components may also lead to changes in failure modes (e.g. modern instrumentation contains microprocessors that have different failure modes from those of their older components).

(b) Changes will occur in standards and regulations, which may necessitate changes in hardware or software and may interfere with the operation of the reactor.

3.6. Operating personnel need to have current knowledge of the operational systems and their associated documentation; therefore, ongoing training is necessary for personnel at all levels. Section 4 of Ref. [4] provides recommendations on training programmes and qualifications for research reactor personnel.

3.7. Evaluation of the cumulative effects of both physical ageing and obsolescence on the safety at a research reactor is an ongoing process, and is assessed in periodic safety reviews or equivalent systematic safety reassessment programmes (e.g. peer reviews); see paras 7.4–7.7 of this Safety Guide.

3.8. In practice, the ageing management programme at a research reactor is accomplished by coordinating existing programmes, including maintenance, periodic testing and inspection programmes (see Ref. [7]), as well as applying good operational practices, research and development (of material behaviour, radiation effects, chemistry, etc.), and incorporating lessons learned from operating experience.

3.9. An effective ageing management programme will contribute to ensuring the availability of the basic safety functions by:

- (a) Shutting down the reactor and maintaining it in a safe shut down state for all operational states or design basis accidents, including, for example, ensuring the availability of the emergency control room;
- (b) Providing for adequate removal of heat after shutdown, in particular from the core, including for design basis accidents;
- (c) Confining radioactive material in order to prevent or mitigate its unplanned release to the environment.

3.10. In an effective ageing management programme, it is recognized that safety documents (such as the safety analysis report, operational limits and conditions, operating procedures and the emergency plan) may also become outdated or even obsolete. Periodic updating of such documents is needed to maintain their conformity with the actual status of the reactor facility and to take into account

feedback from operating experience. Updating will also be necessary when modification⁵ of existing (or installation of new) experimental devices⁶ is being introduced.

AGEING AND DEFENCE IN DEPTH

3.11. Fundamental Safety Principles [8] establishes principles for ensuring protection and safety of workers, the public and the environment. The activities covered by this Safety Guide are directly related to the defence in depth concept⁷, which is associated with Principle 8 on prevention of accidents: "All practical efforts must be made to prevent and mitigate nuclear or radiation accidents."

3.12. Physical ageing of SSCs may increase the probability of common cause failures (i.e. the simultaneous degradation of physical barriers and redundant components), which could result in the impairment of one or more levels of protection provided by the defence in depth concept. Therefore, in the screening process of SSCs for ageing management, no account should be taken of component redundancy or diversity, on the basis that ageing degradation should be regarded as a common cause failure mechanism and that diversity may not provide protection against all potential failures caused by common ageing mechanisms.

3.13. Defence in depth is achieved by multiple physical or functional barriers. The ageing process may affect defence in depth through the failure of one or more of these barriers (e.g. because of corrosion of the cladding material of fuel elements, corrosion of the reactor pool liner or corrosion of piping for primary cooling). An

⁵ Modification is the deliberate changing of, or addition to, an existing reactor configuration, with possible implications for safety, intended to permit continuation of operation of a reactor. Modifications may affect safety systems, safety related items or systems, procedures, documentation or operating conditions.

⁶ An experimental device is a device installed in or around a reactor to utilize the neutron flux and ionizing radiation from the reactor for research, development, isotope production or other purposes.

⁷ Defence in depth is a hierarchical deployment of different levels of diverse equipment and procedures to prevent the escalation of anticipated operational occurrences and to maintain the effectiveness of physical barriers placed between a radiation source or radioactive material and workers, members of the public or the environment, in operational states and, for some barriers, under accident conditions [2].

effective ageing management programme should provide for maintaining defence in depth through:

- (a) Incorporation of good design and engineering features providing safety margins, mainly by use of design, technology and materials of high quality and reliability;
- (b) Compliance with the requirements of the operational limits and conditions;
- (c) Execution of relevant tasks and activities, in accordance with approved operating procedures and by following good practices.

Recommendations on establishing operational limits and conditions, including those relating to ageing, and on establishing operating procedures for research reactors are provided in Section 3 of Ref. [9].

SERVICE CONDITIONS AND AGEING

3.14. In the ageing management programme, account should be taken of the fact that service conditions are the major contributors to ageing of SSCs, through chemical and physical processes that affect material properties or functional capabilities. Service conditions include:

- (a) Stress and/or strain;
- (b) Temperature;
- (c) Pressure;
- (d) The chemistry regime;
- (e) Environmental factors such as radiation, high humidity, or the presence and use of chemically active liquids and gases;
- (f) Service wear, including changes in dimensions and/or the relative positions of individual parts of assemblies.

3.15. Limiting values for service conditions (e.g. the maximum temperature of fuel cladding, the pressure in cooling lines and across filters, the vibration levels of primary cooling pumps, the content of non-dissolved solids in water coolant, and the water coolant chemistry parameters such as electrical conductivity and pH) should be included in the limiting conditions for safe operation⁸ of the

⁸ Limiting conditions for safe operation are established constraints on equipment and operational parameters that must be adhered to during startup, operation, shutting down and shutdown to provide an acceptable assurance of safety at a research reactor according to the licence conditions and within applicable regulations.

operational limits and conditions. Further recommendations on the factors to be considered in establishing limiting conditions for safe operation of a research reactor are provided in Appendix I of Ref. [9].

3.16. There are three main categories of service conditions that should be considered in the ageing management programme. These categories are:

- (1) Service conditions associated with normal operation;
- (2) Service conditions associated with anticipated operational occurrences;
- (3) Environmental conditions.

Service conditions lead to degradation of SSC materials through, for example, one or more of the following ageing mechanisms⁹:

- (a) A change in physical proprieties (e.g. swelling, chemical decomposition, and changes in material strength, ductility or resistivity);
- (b) Irradiation and thermal embrittlement;
- (c) Creep;
- (d) Fatigue, including thermal fatigue;
- (e) Corrosion, including galvanic corrosion, corrosion erosion and corrosion assisted cracking;
- (f) Wear (e.g. fretting) and wear assisted cracking (e.g. fretting fatigue).

Further information on the categories of service conditions and the associated ageing mechanisms can be found in Section 3 of Ref. [10]. Annex I of the present Safety Guide describes the effect of ageing for different categories of service conditions.

⁹ An ageing mechanism is a specific process that gradually changes the characteristics of an SSC with time or use.

4. AGEING CONSIDERATIONS IN DIFFERENT STAGES OF THE LIFETIME OF A RESEARCH REACTOR

GENERAL

4.1. Ageing management of SSCs important to safety should be implemented proactively (i.e. with foresight and anticipation) throughout the lifetime of the research reactor, i.e. in design, fabrication and construction, commissioning, operation (including utilization and modification) and decommissioning. Particular consideration should be given to ageing management for periods of extended shutdown. Reference [1] requires that "Ageing effects shall be taken into account for all operational states, including periods of maintenance and shutdown."

4.2. The operating organization should be able to demonstrate that regulatory requirements for ageing management are met and that relevant issues specific to the ageing of a research reactor are clearly identified and documented in the safety analysis report and supporting documents throughout the lifetime of the reactor. The safety analysis report should include (or should reference documents that contain) information that confirms that all materials of SSCs important to safety have been selected to withstand the service conditions to which they are subjected, without unacceptable degradation of the performance of their intended safety functions. The safety analysis report should also describe the conduct of the in-service inspection programme, and should provide information on SSCs to be inspected, as well as their inspection frequency and criteria. Further guidance on the preparation of safety analysis reports for research reactors is provided in the Appendix of Ref. [11].

4.3. Feedback from relevant experience at other reactors, including nuclear power plants, should be used by the operating organization in evaluating the ageing management measures proposed by designers and suppliers.

4.4. An independent review of the ageing management programme and any subsequent modifications to this programme at various stages throughout the lifetime of the research reactor should be performed by the safety committee or other competent body. Any proposed modifications to the ageing management programme should be submitted to the regulatory body for review and assessment, if so required by national regulations. Paragraphs 3.5–3.9 of Ref. [12] provide further information on categorization of research reactor modifications.

DESIGN

4.5. For the design of a research reactor, including modifications and new experiments, the following requirements apply in respect of selection and ageing of materials:

"At the design stage, an appropriate safety margin shall be adopted to allow for the anticipated properties of materials at the end of their useful lifetime" (para. 6.68, Ref. [1]).

"To ensure the capability of all items important to safety to perform their safety functions, appropriate margins shall be provided in the design to take account of relevant ageing effects and potential ageing related degradation" (para. 6.69, Ref. [1]).

"Provisions shall also be made for the necessary monitoring, testing, sampling and inspection for the detection, assessment, prevention and mitigation of ageing effects" (para. 6.70, Ref. [1]).

4.6. In the design of, and any modification to, a research reactor, including installation of new experimental devices or changes to existing equipment, consideration should be given to the needs of the operating organization with regard to ageing management and reactor utilization [12].

4.7. The design of, and any modification to, a research reactor or experimental device should facilitate inspections and testing aimed at detecting ageing mechanisms and their degrading effects on SSCs, while maintaining the principle that radiation exposure of inspection personnel should be kept as low as reasonably achievable. Particular consideration should be given to SSCs that are difficult to inspect or maintain. Further recommendations on design for inspectability can be found in paras 4.17 and 4.18 of Ref. [7].

4.8. The operating organization should document the ageing management programme and should thereby be able to demonstrate that ageing issues have been adequately addressed in the design for the entire lifetime of the research reactor.

4.9. In procurement documents for SSCs, the operating organization should specify requirements to facilitate ageing management, including information to be included in documents received from suppliers.

4.10. In the reactor design:

- (a) It should be ensured that ageing effects are taken into account in equipment qualification¹⁰ activities, with consideration given to all the conditions established in the design basis (including anticipated operational occurrences and design basis accidents).
- (b) All potential ageing mechanisms for passive and active¹¹ SSCs that could affect the ability to perform the safety functions of the SSCs during their design life should be identified, evaluated and taken into account. In doing this, the importance of the SSCs to safety should be identified, and account should be taken of the particular service conditions, required mode of operation, expected service life and ease of replacement.
- (c) Relevant ageing management experience from nuclear installations (including other research reactors and nuclear power plants) and other industrial installations, as well as research results, should be reviewed and taken into account.
- (d) Consideration should be given to the use of advanced materials (and their production processes) with greater ageing resistance properties (e.g. materials with a high resistance to corrosion, or high strength); consideration should also be given to the combined effects of several ageing mechanisms acting simultaneously.
- (e) Consideration should be given to maintenance requirements and the need for material testing programmes, including surveillance specimens, to monitor ageing degradation.
- (f) Consideration should be given to the use of compatible materials, especially materials used for welding.

4.11. Ageing management should be included as a topic in the general design criteria and should address the following points:

(a) Different types of SSCs important to safety (concrete structures, mechanical components and equipment, electrical and instrumentation, and control equipment and cables, etc.) and measures to monitor their potential degradation;

¹⁰ Equipment qualification is the generation and maintenance of evidence to ensure that equipment will operate on demand, under specified service conditions, to meet system performance requirements [2].

¹¹ A passive component is a component whose functioning does not depend on an external input such as actuation, mechanical movement or supply of power. Any component that is not a passive component is an active component [2].

- (b) General principles stating how the environment of SSCs is to be maintained within specified service conditions (location of ventilation, insulation of hot SSCs, radiation shielding, damping of vibrations, avoidance of submerged conditions and selection of cable routes);
- (c) Proposals for appropriate monitoring and sampling programmes for materials in cases where it is found that ageing degradation may occur that may affect the capability of SSCs to perform their safety function throughout the lifetime of the reactor.

FABRICATION AND CONSTRUCTION

- 4.12. The operating organization should ensure that:
- (a) Relevant information on factors affecting ageing management, including service conditions, is provided to SSC manufacturers, and, through audits of the manufacturers' management system, this information is properly taken into account in the fabrication and construction of SSCs.
- (b) Current knowledge about relevant ageing mechanisms and degradation as well as possible mitigation measures is taken into account in the fabrication and construction of SSCs.
- (c) Reference (baseline) data, including manufacturing and inspection records, as well as records on shipment and storage conditions, are collected and documented.
- (d) Surveillance specimens for specific ageing monitoring programmes are made available and installed in accordance with design specifications.

COMMISSIONING

4.13. The operating organization should establish a systematic programme for measuring and recording baseline data relevant to ageing management for SSCs important to safety. This includes mapping the actual environmental conditions at each critical spot of the reactor to ensure that they are in compliance with the design specifications.

4.14. Special attention should be paid to identification of hot spots in terms of temperature and dose rate, to measurement of vibration levels, and to characterization of thermal insulation or electrical isolation. All parameters that can influence ageing degradation should be identified in commissioning and should be tracked throughout the reactor lifetime (see also paras 7.4–7.6 and

A.20(p) of Ref. [13]). Acceptance criteria for these parameters should also be established.

OPERATION

4.15. The following issues should be taken into account in implementing a systematic ageing management programme:

- (a) Support for the ageing management programme by the management of the operating organization;
- (b) Early implementation of an ageing management programme;
- (c) A proactive approach based on an adequate understanding of SSC ageing, rather than a reactive approach responding to SSC failures;
- (d) Optimal operation of SSCs to slow down the rate of ageing degradation;
- (e) Proper implementation of maintenance and testing activities in accordance with operational limits and conditions, design requirements and manufacturers' recommendations, and following approved operating procedures;
- (f) Minimization of human performance factors that may lead to premature degradation, through enhancement of staff motivation, sense of ownership and awareness, and understanding of the basic concepts of ageing management;
- (g) Availability and use of correct operating procedures, tools and materials, and of a sufficient number of qualified staff for a given task;
- (h) Appropriate storage of spare parts and consumables susceptible to ageing, to minimize degradation while in storage and to control their shelf life properly;
- (i) Availability of the necessary competences for dealing with complex ageing issues;
- (j) Effective internal and external communication;
- (k) Feedback of operating experience (both generic and plant specific operating experience, including operating experience from non-nuclear plants), to learn from relevant ageing related events (see paras 2.15 and 2.16 of Ref. [1]);
- (l) Follow-up of possible degradation trends in SSCs between successive periodic testing;
- (m) Use of databases on SSC reliability and maintenance histories;
- (n) Use of adequate and qualified methods of non-destructive testing and ageing monitoring for early detection of flaws possibly resulting from intensive use of equipment.

4.16. The operating organization should identify and address the following potentially significant common weaknesses of ageing management:

- (a) Insufficient understanding and predictability of ageing at the time of design and construction of a research reactor;
- (b) Premature ageing of SSCs in research reactors (i.e. ageing degradation that occurs earlier than expected) caused by non-compliance with the requirements of the management system, unforeseen ageing phenomena, or operation of SSCs beyond their design specifications or in service conditions more severe than the normal environmental conditions.

4.17. The operating organization should identify and justify possible changes in operational conditions (e.g. radiation levels, flow distribution, coolant velocity and vibration) that could cause accelerated or premature ageing and failure of some SSCs, in the event of:

- Uprating of the reactor power¹²;
- Installation of new experimental devices or changes in the utilization programme;
- Implementation of modifications, including changes to safety analysis;
- Replacement of SSCs.

4.18. If a new ageing mechanism is discovered (e.g. through feedback from operating experience or research), the operating organization should perform an appropriate review of the ageing management programme to determine the effects, if any, on SSCs important to safety which are operating at similar conditions; the operating organization should put in place appropriate ageing management measures.

4.19. For SSCs important to safety, the operating organization should consider preparing contingency maintenance plans to deal with the potential degradation or failure of these SSCs caused by potential ageing mechanisms and effects.

¹² Examples of significant ageing effects due to power uprating include radiation embrittlement of the reactor vessel, flow accelerated corrosion and vibration of primary system piping.

UTILIZATION AND MODIFICATIONS

4.20. Research reactors are frequently modified to incorporate new experimental devices or to improve the efficiency of normal operation. This and the fact that the frequency of testing and repair may also need to be changed due to ageing or obsolescence of SSCs are particularly important for the maintenance programme (which, in practice, forms the basis for the ageing management programme). Consideration should be given to reactor modifications, including temporary ones, with regard to the possible acceleration of ageing phenomena or the appearance of phenomena that have not previously been considered.

4.21. Consideration should be given to the proper cooling of experimental facilities and reactor structures such as thermal columns and concrete shields, as well as to electrical and instrumentation cables, which may be located in unventilated hot areas.

4.22. Particular consideration should be given to:

- (a) Experimental devices operating at high temperature gradients, high temperatures and/or pressures causing creep and fatigue of materials, and beam tubes;
- (b) Degradation of mechanical properties of experimental devices due to radiation induced embrittlement, or due to vibration induced by water flow;
- (c) Irradiation of capsules containing materials (e.g. copper or mercury) that can cause corrosion in aluminium alloys.

4.23. The documentation for a research reactor may become obsolete due to changes in the reactor utilization programme (e.g. due to installation of a new experimental device or changes to an existing experiment) or modifications. Requirements on updating of operating procedures, drawings and other technical documentation should be included in the ageing management programme.

EXTENDED SHUTDOWN

4.24. Reference [1] requires that "The operating organization shall take appropriate measures during an extended shutdown to ensure that materials and components do not seriously degrade." The following measures should be implemented:
- (a) Unloading of the fuel elements from the reactor core to storage racks as appropriate;
- (b) Removing components for protective storage;
- (c) Maintaining water chemistry within specifications and maintaining appropriate environmental conditions (e.g. humidity and temperature) to prevent accelerated ageing (e.g. corrosion).

Further information on safety considerations for research reactors in extended shutdown is provided in Ref. [14].

4.25. Periods of extended shutdown should be used for additional inspections that are not included in the operational limits and conditions for SSCs during normal operation (e.g. because the core cannot be unloaded or the beryllium reflector cannot be dismantled). Such additional inspections include:

- Inspection of the reactor vessel or core support structure;
- Inspection of the core structure internals, including connection bolts;
- Dimensional control and alignment of the core internals and reactivity control mechanisms¹³;
- Inspection of bellows, strainers, valves and other piping of SSCs important to safety;
- Inspection of beam ports.

4.26. Particular consideration should be given to inspections for degradation of civil structures (including biological shielding) and mechanical and electrical SSCs. Documentation and software should also be evaluated for obsolescence.

DECOMMISSIONING

4.27. Appropriate arrangements should be made to ensure that the required equipment and SSCs (e.g. means of confinement, radiation monitoring, long term cooling, lifting equipment and condition monitoring equipment) remain available and functional, to facilitate decommissioning activities.

¹³ Reactivity control mechanisms are devices for controlling the reactivity, including regulating rods, control rods, shutdown rods and blades, and devices for controlling the moderator level.

5. ELEMENTS OF AGEING MANAGEMENT PROGRAMMES

GENERAL

5.1. Ageing management programmes for research reactors should be aimed at identification and implementation of effective and appropriate ageing management actions and practices that provide for timely detection and mitigation of ageing effects in SSCs.

5.2. A systematic ageing management programme for the research reactor should be applied, comprising the following elements:

- Screening of SSCs for ageing management review;
- Identification and understanding of ageing degradation;
- Minimization of ageing degradation;
- Detection, monitoring and trending of ageing degradation;
- Mitigation of ageing degradation;
- Continuous improvement of the ageing management programme;
- Record keeping.

5.3. The ageing management programme should be applied at all stages during the lifetime of the research reactor, including the design stage.

SCREENING OF SSCs FOR AGEING MANAGEMENT REVIEW

5.4. A research reactor has a large number and variety of SSCs, some of which are more important to safety than others. There are also considerable differences in the extent to which these SSCs are susceptible to ageing degradation. It is neither practicable nor necessary to evaluate and quantify the extent of ageing degradation in every individual SSC. A systematic approach should therefore be applied to focus resources on those SSCs, including experimental devices, that can have a negative impact on the safe operation of a reactor and that are susceptible to ageing degradation.

5.5. The safety based approach outlined in the following should be applied to the screening of SSCs for a review of the management of ageing:

- (1) First level screening: The existing list of SSCs important to safety, which should have been developed during the design stage, should be reviewed for completeness (Annex II). If such a list is not available, first level screening should be performed to identify those SSCs that are important to safety, on the basis of whether or not a failure¹⁴ could lead (directly or indirectly) to the loss or impairment of a safety function.
- (2) Second level screening: For each of the SSCs identified as being important to safety, the specific structural elements¹⁵ and components that are important to safety should be identified, i.e. those whose failure could lead (directly or indirectly) to the loss or impairment of a safety function.
- (3) Third level screening: From the list of structural elements and components important to safety, those for which ageing degradation has the potential to cause component failure should be identified, and a justification should be provided for the components that are excluded.

An outline of this screening process is given in Fig. 1.

5.6. Alternative screening methodologies (e.g. approaches based on probabilistic safety assessment) may also be acceptable provided that they are carried out in accordance with the established requirements for research reactor safety. The specific screening methodology used should be justified and documented.

5.7. Annex II provides a sample list of SSCs that need to be considered in the ageing management programme of open pool, light water research reactors and heavy water moderated research reactors. It is emphasized that the ageing management programme for a specific research reactor need not be limited to the SSCs presented in Annex II. The SSCs specific to the given research reactor should also be considered.

5.8. In the ageing management programme, consideration should be given to SSCs important to safety for normal operation and design basis accidents. These SSCs should be identified by applying the approach set out in para. 5.5. For efficiency, consideration should be given to grouping similar components (e.g. valves, pumps and small diameter piping) that operate in comparable service conditions (e.g. pressure, temperature and water chemistry).

¹⁴ Failure is the inability of an SSC to function within acceptance criteria [2].

¹⁵ Structural elements include both simple structures and elements of complex structures. The term 'structural element' is used only for this screening process. In subsequent paragraphs, structural elements are again referred to as 'structures'.



FIG. 1. Outline of the process for screening SSCs for ageing management.

IDENTIFICATION AND UNDERSTANDING OF AGEING DEGRADATION

5.9. To understand the ageing degradation of a structure or component, its ageing mechanisms and effects should be identified and understood; understanding ageing is the basis for the effective monitoring and mitigation of ageing effects.

5.10. Interactions between materials, and service conditions, should be assessed to identify degradation mechanisms. This assessment should be performed using available analytical or empirical models for evaluating past ageing degradation and predicting future ageing degradation.

5.11. Existing methods for inspection, testing, surveillance, monitoring and assessment should be evaluated, with account taken of relevant operating experience and research results, to determine whether they are effective for timely detection of ageing degradation before failure of the structure or component occurs.

MINIMIZATION OF AGEING DEGRADATION

5.12. In order to limit the effects of ageing degradation, preventive actions should be taken. Ideally, preventive actions should be determined during the design stage of the research reactor. Preventive actions should be continuously improved, with account taken of relevant operating experience and research results, and should include:

- (a) An assessment of the effectiveness of current maintenance and restoration methods and practices (including refurbishment and periodic replacement of parts and consumables) to control ageing degradation of components;
- (b) Establishment of appropriate operating conditions and implementation of practices that minimize ageing degradation of components;
- (c) Possible changes in design, materials or environmental conditions, where applicable, to minimize ageing degradation of components.

5.13. Periodic reviews should be undertaken at appropriate intervals to determine the effectiveness of actions and practices for preventing ageing degradation of SSCs. Information obtained in periodic reviews should be documented. If the effectiveness of the preventive actions in place is determined not to be acceptable, appropriate corrective actions should be implemented.

DETECTION, MONITORING AND TRENDING OF AGEING DEGRADATION

5.14. An assessment of various examination methods should be performed to select those methods that are appropriate for detecting effects of the identified ageing mechanisms. Such examination methods should include:

- Inspections;
- Monitoring;
- Performance tests;
- Periodic testing;
- Non-destructive testing, including visual examination.

5.15. The results of these examinations should be evaluated using baseline data collected in previous examinations, to determine whether the conditions¹⁶ of the SSCs are acceptable for continued safe operation or whether remedial measures need to be taken. The examination results should be added to the baseline data for use in subsequent examinations, and records should be kept of any remedial measures.

5.16. The capability and feasibility of detection methods should be periodically evaluated to check their adequacy (including whether they are sufficiently sensitive, reliable and accurate).

5.17. The frequency of examination of an SSC should be in accordance with the recommendations of the designer and/or manufacturer. The frequency may be adjusted on the basis of the likelihood of failure of the SSC and on the basis of experience, including experience acquired from similar facilities. Any proposed changes to the frequency of examinations should be justified.

5.18. Inspection activities should include:

- (a) Observation of the condition of SSCs (e.g. leaks, noise and vibration); this is normally done during periodic walkdowns of the research reactor.
- (b) Sampling of water coolant for chemical¹⁷ or radiochemical analysis.

5.19. Ageing effects may be detected by a change in measurable operating parameters (e.g. control rod drop time, water chemistry parameters, temperature, flow rate and pressure). Parameters that can be predictive of ageing degradation should be routinely monitored (either on-line or periodically). Readings should

¹⁶ Condition assessment of an SSC is an assessment to determine its current performance and condition (including assessment of any ageing related failures or indications of significant material degradation), and to predict its expected future performance, future ageing degradation and residual service life.

¹⁷ Use of electrochemical techniques for on-line monitoring of corrosion is also suitable for low temperature environments such as research reactor pools.

be assessed and trends determined, in order to predict the onset of ageing degradation in a timely manner.

5.20. Ageing effects can be detected by checking the performance of SSCs. The results of performance tests undertaken as part of the maintenance programme should be periodically examined for evidence of trends that indicate ageing degradation.

5.21. The results of periodic testing performed to verify compliance with the operational limits and conditions should be evaluated to detect and correct abnormal conditions before they give rise to significant consequences for safety.

5.22. Non-destructive testing methods should also be used to detect ageing degradation.¹⁸ Various non-destructive testing methods and their capabilities, together with the associated techniques, are given in Table 1. Further information on non-destructive testing methods and techniques that can be used for detecting ageing of SSCs in research reactors can be found in Ref. [15].

MITIGATION OF AGEING DEGRADATION

5.23. The effectiveness of existing methods and practices for mitigating ageing degradation of a component should be evaluated, with account taken of relevant operating experience and research results. Methods and practices to be evaluated

Method	Type of defect	Techniques		
Visual examination	Scratches, wear, cracks, corrosion or erosion of surfaces	Television cameras, underwater telescopes, binoculars and mirrors		
Surface examination	Surface or near surface flaws or discontinuities	Liquid penetrant and magnetic particles		
Volumetric examination	Depth or size of a surface breaking or subsurface flaw	Radiographic, ultrasonic and eddy current techniques		

TABLE 1. NON-DESTRUCTIVE TESTING METHODS AND ASSOCIATED TECHNIQUES

 $^{^{18}\,}$ For removable SSCs, destructive tests may be envisaged if the results of non-destructive tests have indicated abnormal values.

should include maintenance, refurbishment and periodic replacement of components, modification of SSCs, and altering of operating conditions and practices that may affect the rate of ageing degradation of components.

5.24. It should be noted that while mitigatory and preventive actions are conceptually different, some preventive actions (e.g. maintaining water chemistry parameters within specifications) are also mitigatory actions.

5.25. Once ageing degradation has been detected or predicted, methods for mitigation of ageing effects should be evaluated and mitigatory actions should be taken.

CONTINUOUS IMPROVEMENT OF THE AGEING MANAGEMENT PROGRAMME

5.26. The management of the research reactor should establish provisions for performance review and continuous improvement of the ageing management programme. Ageing evaluations and condition assessments should be periodically reviewed and updated, and the effectiveness of ageing management activities for individual SSCs should be periodically evaluated in the light of current knowledge and adjusted as appropriate. Current knowledge should be acquired from information on the operation of SSCs, surveillance and maintenance records, lessons learned from other similar facilities and information from the results of research. Consideration should be given to the fact that new ageing degradation mechanisms or unexpected ageing degradation locations may be identified as a consequence of analysis of an incident occurring at another research reactor.

5.27. The following actions should be taken for continuous improvement of ageing management programmes:

- (a) At least once a year, a review should be carried out of the performance of SSCs, indicating the effectiveness of the ageing management programme in monitoring ageing degradation trends and identifying any weaknesses and opportunities for improvement;
- (b) Use of the results of the performance reviews.

5.28. The effectiveness of the ageing management programme should be reviewed periodically, in accordance with the established management system

and in the light of current knowledge and experience. The ageing management programme should be updated as appropriate.

5.29. The result of reviews of the ageing management programme and of assessments of the SSCs, including inspection results and proposed updates, should be submitted to the reactor safety committee (or another competent body) for review and, if required, to the regulatory body for review and assessment.

5.30. Consideration should be given to arranging for peer reviews of the ageing management programme, to obtain an independent assessment, in order to establish whether the ageing management programme is consistent with accepted practices and to identify areas for improvement (see also paras 7.4–7.7 of this Safety Guide).

RECORD KEEPING

5.31. The operating organization should establish a data collection and record keeping system, which should be defined in the safety analysis report (see also paras 2.15 and 2.16 of Ref. [1]), in order to provide information for the following ageing management activities:

- (a) Identification and evaluation of degradation, failures and malfunctions of SSCs caused by ageing effects;
- (b) Prediction of the future performance of SSCs, which is required for continued safe and reliable operation of the research reactor;
- (c) Decisions on the type and timing of preventive maintenance actions, including calibration, repair, modification, refurbishment and replacement, and decisions on adaptations of ageing management programmes;
- (d) Optimization of operating conditions and practices that reduce ageing degradation;
- (e) Identification of new emerging ageing effects before they jeopardize the safety, reliability and service life of the research reactor.

5.32. The data required in relation to ageing management may be divided into the following three categories (see Annex II of Ref. [3] for examples of relevant data):

(1) Baseline information, consisting of data on the design of the reactor and/or of the SSCs and on the conditions at the beginning of the service life of the SSCs;

- (2) Reactor operating records, covering service conditions of the SSCs, including transient data, and data on SSC availability, testing and failure;
- (3) Maintenance records, including information on the condition of the SSCs.

5.33. All records should be retained as specified in the operational limits and conditions.

6. MANAGEMENT OF OBSOLESCENCE

6.1. This section addresses obsolescence (non-physical ageing), which occurs when SSCs become out of date in comparison with current technology, knowledge, standards and regulations or when documentation becomes out of date.

6.2. Obsolescence of SSCs should be identified, and corrective actions should be taken before the occurrence of any decline in reliability or availability of a research reactor.

6.3. Obsolescence of SSCs important to safety should be managed proactively (i.e. with foresight and anticipation) throughout their service life.

6.4. The operating organization should establish activities for the management of obsolescence throughout the lifetime of the research reactor. These activities should be reviewed by the safety committee (or other competent body) and should be submitted to the regulatory body for review and assessment, if required by national regulations. Conditions that can lead to obsolescence and that can thus affect reactor safety include:

- (a) Changes in technology;
- (b) Changes in regulations;
- (c) Documentation becoming out of date (e.g. if it is not correctly updated following modifications of SSCs or changes in the utilization programme).

6.5. Table 2 summarizes these conditions and their effects, and suggests actions that should be considered for managing obsolescence.

6.6. Attention should be paid to the management of obsolescence of technology (which, in general, occurs in systems such as electrical and instrumentation systems, and the radiation monitoring system).

6.7. It should be noted that some of the activities relating to the management of obsolescence will require review and updating of the documentation for the reactor (the safety analysis report, operational limits and conditions, emergency plan, radiation protection programme and operating procedures).

6.8. The documentation for a research reactor should be updated when changes in regulations and standards occur, to prevent its obsolescence.

TABLE 2. TYPES OF OBSOLESCENCE AND ASSOCIATED AGEING EFFECTS, AND RECOMMENDED AGEING MANAGEMENT ACTIONS

Conditions	Ageing effects	Management actions
Changes in technology (safety systems)	Incompatibility between old and new equipment Unavailability of suppliers Shortage or lack of spare parts	Ensure systematic identification of useful service life and anticipated obsolescence Prepare a modification project for future replacement of obsolete SSCs Provide spare parts for planned service life or identify alternative suppliers
Changes in standards and regulations, advances in knowledge	Outdated knowledge of practices, standards and regulations Deviations from current standards and regulations	Ensure compliance with current standards and regulations Consider the modification of SSCs important to safety, as required
Documentation becoming out of date	Lack of the information needed for safe operation	Establish an effective integrated management system, including configuration management

7. INTERFACES WITH OTHER TECHNICAL AREAS

INTRODUCTION

7.1. This section addresses the technical areas that interface with, or are closely related to, the ageing management programme and ageing management activities. It provides guidance on the issues that should be considered, to enhance an ageing management programme, in implementing the activities in these areas. These technical areas are:

- Maintenance, periodic testing and inspection;
- Periodic safety reviews;
- Equipment qualification;
- Reconstitution of the design basis;
- Configuration management;
- Continued safe operation;
- Post-service surveillance and testing.

MAINTENANCE, PERIODIC TESTING AND INSPECTION

7.2. The objective of maintenance, periodic testing and inspection programmes is to ensure that the SSCs function in accordance with the design intentions and requirements, as well as in compliance with the safety analysis report and the operational limits and conditions. The activities in these programmes are closely related to those that should be established in the framework of an ageing management programme in the sense that:

- (a) Preventive maintenance consists of regularly scheduled inspections, testing, servicing, overhauls and replacement activities, which are established inter alia to detect and prevent incipient failures, and to ensure the continuing capability of SSCs to perform their intended functions.
- (b) Corrective maintenance consists of repair and replacement activities not occurring on a regular schedule.
- (c) Periodic testing is carried out to ensure compliance with the operational limits and conditions.
- (d) Inspection, while an activity inherent to all maintenance and periodic testing, is an examination of SSCs for ageing degradation, to determine whether they are acceptable for continued safe operation or whether remedial measures should be taken. Inspection also refers to non-routine

examination for assessment of ageing effects. Inspection is sometimes referred to as in-service inspection.

Further guidance on the interrelationship between maintenance, periodic testing and inspection, and their relation with ageing management for research reactors, is provided in paras 3.5 and 3.6 of Ref. [7].

7.3. In the maintenance programme, it should be taken into account that the maintenance frequency, testing methods and procedures, and repair methods and procedures may need to be adjusted with time, as inadequate maintenance in conjunction with service conditions, as well as improper maintenance, may accelerate ageing degradation of SSCs. The maintenance programme should be evaluated and, if considered necessary, updated on the basis of the findings of the ageing management programme.

PERIODIC SAFETY REVIEW

7.4. In Ref. [8], the text accompanying Principle 3 on leadership and management states that "For operations that continue over long periods of time, assessments are reviewed and repeated as necessary. Continuation of such operations is subject to these reassessments demonstrating to the satisfaction of the regulatory body that the safety measures remain adequate." Concerning such safety reassessment, the operating organization should carry out periodic safety reviews, including reviews of modifications, changes in utilization and ageing management.

7.5. In the framework of the periodic safety review, the operating organization should assess the effects of ageing on the safety of the research reactor, the effectiveness of the ageing management programme and the need for improvements.

7.6. Outcomes of the review of ageing management within the periodic safety review should be used:

- (a) To determine whether the research reactor or individual SSCs important to safety can be operated safely for a specified future period (e.g. the period between the current safety review and the next one);
- (b) To provide inputs for improvement of the scope, frequency and procedures for maintenance, surveillance and inspection, for updating of the safety

analysis, and for modifications of operating conditions or design (including possible changes to the design basis of any SSCs).

Further recommendations and information on periodic safety review for research reactors are provided in paras 2.65–2.68 of Ref. [4].

7.7. In performing a periodic safety review, self-assessment or audits by the operating organization, or peer reviews, the information provided in Ref. [16] should be taken into account. This publication covers the preparation, execution, reporting and follow-up of safety reviews of research reactors, and is applicable inter alia for reassessment of research reactor safety, in order to monitor continued adherence to good safety practices, and for long term safety reviews when ageing of a research reactor may be a point of concern.

EQUIPMENT QUALIFICATION

7.8. Equipment qualification activities should provide for minimization of ageing degradation, through the surveillance, maintenance and periodic replacement of installed equipment or age sensitive parts. The ageing of individual items of equipment should be managed by using the concept either of 'qualified life'¹⁹ or of 'qualified condition'²⁰ established by equipment qualification (see also paras 7.5–7.8 of Ref. [3]).

RECONSTITUTION OF THE DESIGN BASIS FOR SSCs

7.9. The design basis for SSCs is the information that identifies the specific functions to be performed and the controlling design parameters and specific values, or ranges of values, for these parameters.

7.10. The design basis should be reconstituted if it has not been supplied in full to the operating organization, or if it is not adequately documented. This is

¹⁹ The qualified life is the period for which an SSC has been demonstrated, through testing, analysis or experience, to be capable of functioning within acceptance criteria during specific operating conditions while retaining the ability to perform its safety functions in a design basis accident or earthquake [2].

²⁰ The qualified condition is the condition of equipment expressed in terms of one or more measurable condition indicators for which it has been demonstrated that the equipment will meet its performance requirements.

particularly important in the case of installation of a new experiment with major safety significance, or where SSCs have been significantly modified and/or may become vulnerable to ageing degradation. The ageing management programme should be used to demonstrate that, in spite of ageing degradation, the design basis for SSCs important to safety remains valid.

CONFIGURATION MANAGEMENT

7.11. Configuration management is the process of identifying and documenting the characteristics of a research reactor's SSCs [2]. The operating organization, within its management system, should establish a configuration management process aimed at ensuring that changes to characteristics of SSCs due to ageing are properly assessed, recorded and incorporated into the documentation of the research reactor.

CONTINUED SAFE OPERATION

7.12. With proper implementation of the ageing management programme, the lifetime of a research reactor may be extended. The lifetime of a research reactor is usually determined by the reactor's continued usefulness and by its operational costs. In order to assess the continued safe operation of a research reactor, data from the ageing management programme should be analysed and the following actions should be carried out:

- (a) A safety review of the reactor should be performed, including inspection, to establish the actual status of SSCs.
- (b) An overview of potential refurbishment needs should be developed by establishing a comprehensive list of SSCs.
- (c) SSCs important to safety should be selected and relevant ageing mechanisms should be identified, in order to perform a preliminary evaluation.
- (d) The technical and economic feasibility of a refurbishment programme should be established.

7.13. Before commencing a large scale refurbishment project, a strategic plan (or other equivalent study) should be prepared. In addition to commercial and utilization issues, this plan should consider safety issues such as satisfaction of regulatory requirements as well as the technical status of SSCs and experimental devices that are not part of the refurbishment project.

POST-SERVICE SURVEILLANCE AND TESTING

7.14. After final shutdown of a research reactor and before its decommissioning, a post-service surveillance and testing programme should be applied to detect and assess continuing ageing effects. This programme should continue to be applied as long as particular SSCs of the reactor are required to remain in operation and the decommissioning process has not yet been completed.

REFERENCES

- INTERNATIONAL ATOMIC ENERGY AGENCY, Safety of Research Reactors, IAEA Safety Standards Series No. NS-R-4, IAEA, Vienna (2005).
- [2] INTERNATIONAL ATOMIC ENERGY AGENCY, IAEA Safety Glossary: Terminology Used in Nuclear Safety and Radiation Protection, 2007 Edition, IAEA, Vienna (2007).
- [3] INTERNATIONAL ATOMIC ENERGY AGENCY, Ageing Management for Nuclear Power Plants, IAEA Safety Standards Series No. NS-G-2.12, IAEA, Vienna (2009).
- [4] INTERNATIONAL ATOMIC ENERGY AGENCY, The Operating Organization and the Recruitment, Training and Qualification of Personnel for Research Reactors, IAEA Safety Standards Series No. NS-G-4.5, IAEA, Vienna (2008).
- [5] INTERNATIONAL ATOMIC ENERGY AGENCY, The Management System for Facilities and Activities, IAEA Safety Standards Series No. GS-R-3, IAEA, Vienna (2006).
- [6] INTERNATIONAL ATOMIC ENERGY AGENCY, Application of the Management System for Facilities and Activities, IAEA Safety Standards Series No. GS-G-3.1, IAEA, Vienna (2006).
- [7] INTERNATIONAL ATOMIC ENERGY AGENCY, Maintenance, Periodic Testing and Inspection of Research Reactors, IAEA Safety Standards Series No. NS-G-4.2, IAEA, Vienna (2006).
- [8] EUROPEAN ATOMIC ENERGY COMMUNITY, FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, INTERNATIONAL ATOMIC ENERGY AGENCY. INTERNATIONAL LABOUR ORGANIZATION. INTERNATIONAL MARITIME ORGANIZATION, OECD NUCLEAR ENERGY AGENCY, PAN AMERICAN HEALTH ORGANIZATION, UNITED NATIONS PROGRAMME, WORLD ENVIRONMENT HEALTH ORGANIZATION, Fundamental Safety Principles, IAEA Safety Standards Series No. SF-1, IAEA, Vienna (2006).
- [9] INTERNATIONAL ATOMIC ENERGY AGENCY, Operational Limits and Conditions and Operating Procedures for Research Reactors, IAEA Safety Standards Series No. NS-G-4.4, IAEA, Vienna (2008).
- [10] INTERNATIONAL ATOMIC ENERGY AGENCY, Management of Research Reactor Ageing, IAEA-TECDOC-792, IAEA, Vienna (1995).
- [11] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety Assessment of Research Reactors and Preparation of the Safety Analysis Report, IAEA Safety Series No. 35-G1, IAEA, Vienna (1994).
- [12] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety in the Utilization and Modification of Research Reactors, IAEA Safety Series No. 35-G2, IAEA, Vienna (1994).
- [13] INTERNATIONAL ATOMIC ENERGY AGENCY, Commissioning of Research Reactors, IAEA Safety Standards Series No. NS-G-4.1, IAEA, Vienna (2006).
- [14] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety Considerations for Research Reactors in Extended Shutdown, IAEA-TECDOC-1387, IAEA, Vienna (2004).

- [15] INTERNATIONAL ATOMIC ENERGY AGENCY, Application of Non-Destructive Testing and In-Service Inspection to Research Reactors, IAEA-TECDOC-1263, IAEA, Vienna (2001).
- [16] INTERNATIONAL ATOMIC ENERGY AGENCY, Guidelines for the Review of the Research Reactor Safety, Reference Document for the IAEA Integrated Safety Assessment for Research Reactors (INSARR), IAEA Service Series No. 1, IAEA, Vienna (1998).

Annex I

EFFECT OF AGEING FOR DIFFERENT SERVICE CONDITIONS

I-1. The basic cause of ageing degradation of an SSC is the service conditions, which support the actuation of particular ageing mechanisms and lead, unless ageing is properly managed, to loss of the safety function capability of the SSC. Service conditions can be categorized as normal operation, anticipated operational occurrences and environmental conditions.

I–2. The possible effects of ageing (consequence/failure) for service conditions associated with normal operation, with anticipated operational occurrences and for environmental conditions, with associated ageing mechanisms, are summarized in Tables I–1 to I–3, respectively.

Condition	Ageing mechanism	Consequence/failure
Radiation	Change of properties	Chemical decomposition Change of strength Change of ductility Swelling Change of resistivity Burnup
Temperature	Change of properties	Change of strength Change of resistivity Change of ductility
Stress (pressure)	Creep	Changes of geometry (e.g. a breakage or collapse)
Cycling of temperature, flow and/or load Flow induced vibrations	Motion	Displacement Change of position or set point Loose connections Material damage (cracks)
	Fatigue	Breakage or collapse Deformation Material damage (cracks)
	Wear	Deterioration of surface Change of dimensions

TABLE I–1. POSSIBLE EFFECTS OF AGEING ON SERVICE CONDITIONS ASSOCIATED WITH NORMAL OPERATION

TABLE I–1. POSSIBLE EFFECTS OF AGEING ON SERVICE CONDITIONS ASSOCIATED WITH NORMAL OPERATION (cont.)

Condition	Ageing mechanism	Consequence/failure
Flow	Erosion	Change of strength
Fluids chemistry	Corrosion/galvanic cells	Release of radioactivity Reduction of strength Deposition of particles Short circuits Leakage

TABLE I–2. POSSIBLE EFFECTS OF AGEING ON SELECTED SERVICE CONDITIONS ASSOCIATED WITH ANTICIPATED OPERATIONAL OCCURRENCES

Condition	Ageing mechanism	Consequence/failure
Power excursion	Thermal and/or mechanical stresses	Deterioration of core components Accelerated ageing of core components
Unbalanced control rod positions	Thermal stresses	Reduction of strength Accelerated ageing
Power–flow mismatch (error in fuel loading)	Thermal stresses	Reduction of strength Accelerated ageing
Primary pump failure	Thermal and mechanical stresses	Accelerated ageing
Erroneous maintenance or operation	Mechanical damage and adverse chemical conditions	Deterioration of systems Corrosion Accelerated ageing

TABLE	I–3.	POSSIBLE	EFFECTS	OF	AGEING	ON	SELECTED
ENVIRO	NME	NTAL CONDI	TIONS				

Condition	Ageing mechanism	Consequence/failure
Humidity and salinity	Corrosion	Leakage Release of radioactive material Reduction of strength Short circuits
Chemical agents	Chemical reactions	Corrosion Deterioration of SSCs
Wind, dust and sand	Erosion and deposition	Change of strength Deterioration of surface Malfunction of mechanical and electrical components

Annex II

EXAMPLE OF SCREENING OF RESEARCH REACTOR SSCs FOR AGEING MANAGEMENT PURPOSES

II–1. Table II–1 provides examples of screening, for ageing management, of the SSCs of pool type, light water research reactors, as well as of heavy water moderated research reactors. The table lists SSCs, and indicates their importance to safety and their ease of replacement, as well as the relevant ageing mechanism. The abbreviations used in Table II–1 are explained following the table.

TABLE II–1. TYPICAL RESEARCH REACTOR SSCs TOGETHER WITH THEIR EASE OF REPLACEMENT AND POSSIBLE AGEING MECHANISMS

SSC	Important to safety?	Ease of replacement	Ageing mechanisms					
Pool and reactor internals (for light water moderated reactors)								
Pool structure and vessel	Y	A/B	1, 2, 4, 5, 6					
Core structure	Y	В	1, 4, 5, 6, 7					
Reflector	Y	B/C	1, 4, 5					
Control rods and mechanisms	Y	С	1, 4, 5					
Shielding	Y	С	1, 5					
Beam tubes	Y	B/C	1, 3, 5					
Liner	Y	B/C	1, 3, 5					
Fuel assemblies and storage in reactor pool	Y	С	1, 5					
Reactor internals (for heavy water moder	ated reactors)							
Reactor tank	Y	A/B	1, 2, 5, 6					
Core structure	Y	В	1, 2, 5					
Control rods and mechanisms	Y	С	1, 4, 5					
Fuel assemblies	Y	С	1, 5					
Cooling systems								
Primary	Y	B/C	1, 3, 4, 5, 6, 7					
Pool	Y	A/B	1, 3, 4, 5, 6					
Emergency	Y	B/C	3, 4, 5, 6					
Make-up	N	С	5					

TABLE	II–1.	TYPI	CAL	RESEARCH	RE	ACTO	R SSCs	TC	OGETHER
WITH 7	THEIR	EASE	OF	REPLACEME	NT	AND	POSSIBI	LE	AGEING
MECHA	ANISMS	6 (cont.)							

SSC	Important to safety?	Ease of replacement	Ageing mechanisms
Purification	Y	С	1, 2, 5
Secondary	Ν	С	4, 5, 6, 7
Hot water layer	Y	С	4, 5, 6
Heavy water storage	Y	B/C	5
Collection of heavy water leaks	Y	B/C	5
Cover gas circulation	Y	С	4, 5
Confinment/containment			
Structure	Y	А	2, 3, 4, 5
Biological shield	Y	A/B	1, 2, 3, 4, 5
Ventilation: normal	М	С	2, 5, 6
Ventilation: emergency	Y	B/C	5, 6
Penetrations	Y	С	1, 2, 4, 5
Isolation system	Y	С	4, 5
Stack	Y	B/C	6
Controls and instrumentation			
Shutdown systems	Y	С	4, 5
Protection system	Y	В	4, 5
Control system	Y	С	2, 4, 6
Control console	М	B/C	2,6
Radiation monitoring	Y	С	5
Process systems	М	В	4, 5, 6
Annunciators	Y	С	2, 4, 6
Instrumentation	М	С	1, 2, 4, 6
Cabling	М	B/C	1, 2, 5
Remote shutdown and remote monitoring	Y	С	5
Pneumatic system	М	С	4, 5
Data acquisition	М	С	4, 5
Seismic protection	Y	С	4, 5
Auxiliaries			
Electrical power system	М	В	6
Emergency power system	Y	B/C	5, 6

TABLE	II-1.	TYPI	CAL	RESEARCH	RE	ACTO	R SSCs	TC	OGETHER
WITH	THEIR	EASE	OF	REPLACEME	NT	AND	POSSIBI	LE	AGEING
MECH	ANISMS	S (cont.)							

SSC	Important to safety?	Ease of replacement	Ageing mechanisms
Fire protection	Y	В	5
Lightning protection	М	B/C	5
Flood protection	М	С	5
Communications	М	C/D	5
Cranes	Y	B/C	4, 5
Handling and storage facilities	Y	D	5
Transfer casks and fuel casks	Y	B/C	1, 4, 5
Handling, storage and disposal of radioactive waste	Y	B/C	1, 2, 5, 6
Hot cells	М	В	1, 5, 6, 8
Compressed air	М	С	4, 5, 6
Laboratories	Ν	С	5, 6, 8
Heavy water auxiliaries, including recombination	Y	С	5, 6
Spare parts	М	B/C	8
Experimental facilities			
Cold and hot sources	Y	B/C	1, 2, 3, 4, 5, 6
Shielding	Y	С	1, 5
Rigs and loops	Y	B/C	1, 2, 3, 4, 5, 6
Beam tube lines	Y	С	1, 3, 4, 5
Isotope production and irradiation facilities	М	С	1, 2, 3, 4, 5, 6
Rabbit systems	М	С	1, 5, 6
Thermal columns	М	С	1, 2, 3, 5
Dry irradiation rooms	Ν	С	1, 5
Reactor block			
Reflector	Y	A/B	1, 4, 5
Thermal shield	Y	A/B	1, 2, 5
Biological shield	Y	A/B	1, 2, 4, 5
Reactor block cooling system	Y	В	2, 5

TABLE II–1. TYPICAL RESEARCH REACTOR SSCs TOGETHER WITH THEIR EASE OF REPLACEMENT AND POSSIBLE AGEING MECHANISMS (cont.)

Other factors (non-SSC)	Important to safety?	Ease of replacement	Ageing mechanisms
Documentation			
Design	Μ	B/C	10
Safety analysis report	Y	B/C	9, 10
Operational limits and conditions	Y	B/C	10
Operating procedures	Y	B/C	10
Management system	Y	B/C	9, 10
Reviews and assessment	Y	С	9, 10
Industrial safety	Ν	B/C	8, 9
Licensing	Y	B/C	9

Abbreviations used in Table II–1

Im	portant to safety?	Ease of replacement	Ageing mechanisms
Y:	Yes	A: Very difficult	1: Changes of properties due to neutron irradiation
N:	No	B: Difficult technically or costly	2: Changes of properties due to temperature service conditions
M:	Maybe, depending on specific reactor design and features; see also footnote 1 on page 1.		3: Stress or creep (due to pressure and temperature service conditions)
		C: Normal	4: Motion, fatigue or wear (resulting from cycling of temperature, flow and/or load, or flow induced vibrations)
		D: Readily	5: Corrosion
			6: Chemical processes
			7: Erosion
			8: Changes of technology
			9: Changes of regulations
			10: Obsolescence of documentation

CONTRIBUTORS TO DRAFTING AND REVIEW

Abou Yehia, H.	International Atomic Energy Agency		
Angell, P.	Atomic Energy of Canada Ltd, Canada		
Boogaard, J.	International Atomic Energy Agency		
Böck, H.	Atominstitut der Österreichischen Universitäten, Austria		
Hargitai, T.	KFKI Atomic Energy Research Institute, Hungary		
Inagaki, T.	International Atomic Energy Agency		
Offerein, J.	Nuclear Research and Consultancy Group, Netherlands		
Rive, D.	Institut de radioprotection et de sûreté nucléaire, France		
Shokr, A.M.	International Atomic Energy Agency		
Villa, M.	Atominstitut der Österreichischen Universitäten, Austria		
Wijtsma, F.J.	Nuclear Research and Consultancy Group, Netherlands		
Willers, A.	International Atomic Energy Agency		

BODIES FOR THE ENDORSEMENT OF IAEA SAFETY STANDARDS

An asterisk denotes a corresponding member. Corresponding members receive drafts for comment and other documentation but they do not generally participate in meetings. Two asterisks denote an alternate.

Commission on Safety Standards

Argentina: González, A.J.; Australia: Loy, J.; Belgium: Samain, J.-P.; Brazil: Vinhas, L.A.; Canada: Jammal, R.; China: Liu Hua; Egypt: Barakat, M.; Finland: Laaksonen, J.; France: Lacoste, A.-C. (Chairperson); Germany: Majer, D.; India: Sharma, S.K.; Israel: Levanon, I.; Japan: Fukushima, A.; Korea, Republic of: Choul-Ho Yun; Lithuania: Maksimovas, G.; Pakistan: Rahman, M.S.; Russian Federation: Adamchik, S.; South Africa: Magugumela, M.T.; Spain: Barceló Vernet, J.; Sweden: Larsson, C.M.; Ukraine: Mykolaichuk, O.; United Kingdom: Weightman, M.; United States of America: Virgilio, M.; Vietnam: Le-chi Dung; IAEA: Delattre, D. (Coordinator); Advisory Group on Nuclear Security: Hashmi, J.A.; European Commission: Faross, P.; International Nuclear Safety Group: Meserve, R.; International Commission on Radiological Protection: Holm, L.-E.; OECD Nuclear Energy Agency: Yoshimura, U.; Safety Standards Committee Chairpersons: Brach, E.W. (TRANSSC); Magnusson, S. (RASSC); Pather, T. (WASSC); Vaughan, G.J. (NUSSC).

Nuclear Safety Standards Committee

Algeria: Merrouche, D.; Argentina: Waldman, R.; Australia: Le Cann, G.; Austria: Sholly, S.; Belgium: De Boeck, B.; Brazil: Gromann, A.; *Bulgaria: Gledachev, Y.; Canada: Rzentkowski, G.; China: Jingxi Li; Croatia: Valčić, I.; *Cyprus: Demetriades, P.; Czech Republic: Šváb, M.; Egypt: Ibrahim, M.; Finland: Järvinen, M.-L.; France: Feron, F.; Germany: Wassilew, C.; Ghana: Emi-Reynolds, G.; *Greece: Camarinopoulos, L.; Hungary: Adorján, F.; India: Vaze, K.; Indonesia: Antariksawan, A.; Iran, Islamic Republic of: Asgharizadeh, F.; Israel: Hirshfeld, H.; Italy: Bava, G.; Japan: Kanda, T.; Korea, Republic of: Hyun-Koon Kim; Libyan Arab Jamahiriya: Abuzid, O.; Lithuania: Demčenko, M.; Malaysia: Azlina Mohammed Jais; Mexico: Carrera, A.; Morocco: Soufi, I.; Netherlands: van der Wiel, L.; Pakistan: Habib, M.A.; Poland: Jurkowski, M.; Romania: Biro, L.; Russian Federation: Baranaev, Y.; Slovakia: Uhrik, P.; Slovenia: Vojnovič, D.; South Africa: Leotwane, W.; Spain: Zarzuela, J.; Sweden: Hallman, A.; Switzerland: Flury, P.; Tunisia: Baccouche, S.;

Turkey: Bezdegumeli, U.; Ukraine: Shumkova, N.; United Kingdom: Vaughan, G.J. (Chairperson); United States of America: Mayfield, M.; Uruguay: Nader, A.; European Commission: Vigne, S.; FORATOM: Fourest, B.; IAEA: Feige, G. (Coordinator); International Electrotechnical Commission: Bouard, J.-P.; International Organization for Standardization: Sevestre, B.; OECD Nuclear Energy Agency: Reig, J.; *World Nuclear Association: Borysova, I.

Radiation Safety Standards Committee

*Algeria: Chelbani, S.; Argentina: Massera, G.; Australia: Melbourne, A.; *Austria: Karg, V.; Belgium: van Bladel, L.; Brazil: Rodriguez Rochedo, E.R.; *Bulgaria: Katzarska, L.; Canada: Clement, C.; China: Huating Yang; Croatia: Kralik, I.; *Cuba: Betancourt Hernandez, L.; *Cyprus: Demetriades, P.; Czech Republic: Petrova, K.; Denmark: Øhlenschlæger, M.; Egypt: Hassib, G.M.; Estonia: Lust, M.; Finland: Markkanen, M.; France: Godet, J.-L.; Germany: Helming, M.; Ghana: Amoako, J.; *Greece: Kamenopoulou, V.; Hungary: Koblinger, L.; Iceland: Magnusson, S. (Chairperson); India: Sharma, D.N.; Indonesia: Widodo, S.; Iran, Islamic Republic of: Kardan, M.R.; Ireland: Colgan, T.; Israel: Koch, J.; Italy: Bologna, L.; Japan: Kiryu, Y.; Korea, Republic of: Byung-Soo Lee; *Latvia: Salmins, A.; Libyan Arab Jamahiriya: Busitta, M.; Lithuania: Mastauskas, A.; Malavsia: Hamrah, M.A.; Mexico: Delgado Guardado, J.; Morocco: Tazi, S.; Netherlands: Zuur, C.; Norway: Saxebol, G.; Pakistan: Ali, M.; Paraguay: Romero de Gonzalez, V.; Philippines: Valdezco, E.; Poland: Merta, A.; Portugal: Dias de Oliveira, A.M.; Romania: Rodna, A.; Russian Federation: Savkin, M.; Slovakia: Jurina, V.; Slovenia: Sutej, T.; South Africa: Olivier, J.H.I.; Spain: Amor Calvo, I.; Sweden: Almen, A.; Switzerland: Piller, G.; *Thailand: Suntarapai, P.; Tunisia: Chékir, Z.; Turkey: Okyar, H.B.; Ukraine: Pavlenko, T.; United Kingdom: Robinson, I.; United States of America: Lewis, R.; *Uruguav: Nader, A.; European Commission: Janssens, A.; Food and Agriculture Organization of the United Nations: Byron, D.; IAEA: Boal, T. (Coordinator); International Commission on Radiological Protection: Valentin, J.; International Electrotechnical Commission: Thompson, I.; International Labour Office: Niu, S.; International Organization for Standardization: Rannou, A.; International Source Suppliers and Producers Association: Fasten, W.; OECD Nuclear Energy Agency: Lazo, T.E.; Pan American Health Organization: Jiménez, P.; United Nations Scientific Committee on the Effects of Atomic Radiation: Crick, M.; World Health Organization: Carr, Z.; World Nuclear Association: Saint-Pierre, S.

Transport Safety Standards Committee

Argentina: López Vietri, J.; **Capadona, N.M.; Australia: Sarkar, S.; Austria: Kirchnawy, F.; Belgium: Cottens, E.; Brazil: Xavier, A.M.; Bulgaria: Bakalova, A.; Canada: Régimbald, A.; China: Xiaoqing Li; *Croatia*: Belamarić, N.; *Cuba: Quevedo Garcia, J.R.; *Cyprus: Demetriades, P.; Czech Republic: Ducháček, V.; Denmark: Breddam, K.; Egypt: El-Shinawy, R.M.K.; Finland: Lahkola, A.; France: Landier, D.; Germany: Rein, H.; *Nitsche, F.; **Alter, U.; Ghana: Emi-Reynolds, G.; *Greece: Vogiatzi, S.; Hungary: Sáfár, J.; India: Agarwal, S.P.; Indonesia: Wisnubroto, D.; Iran, Islamic Republic of: Eshraghi, A.; *Emamjomeh, A.; Ireland: Duffy, J.; Israel: Koch, J.; Italy: Trivelloni, S.; **Orsini, A.; Japan: Hanaki, I.; Korea, Republic of: Dae-Hyung Cho; Libyan Arab Jamahiriya: Kekli, A.T.; Lithuania: Statkus, V.; Malaysia: Sobari, M.P.M.; **Husain, Z.A.; Mexico: Bautista Arteaga, D.M.; **Delgado Guardado, J.L.; *Morocco: Allach, A.; Netherlands: Ter Morshuizen, M.; *New Zealand: Ardouin, C.; Norway: Hornkjøl, S.; Pakistan: Rashid, M.; *Paraguay: More Torres, L.E.; Poland: Dziubiak, T.; Portugal: Buxo da Trindade, R.; Russian Federation: Buchelnikov, A.E.; South Africa: Hinrichsen, P.; Spain: Zamora Martin, F.; Sweden: Häggblom, E.; **Svahn, B.; Switzerland: Krietsch, T.; Thailand: Jerachanchai, S.; Turkey: Ertürk, K.; Ukraine: Lopatin, S.; United Kingdom: Sallit, G.; United States of America: Boyle, R.W.; Brach, E.W. (Chairperson); Uruguay: Nader, A.; *Cabral, W.; European Commission: Binet, J.; IAEA: Stewart, J.T. (Coordinator); International Air Transport Association: Brennan, D.; International Civil Aviation Organization: Rooney, K.; International Federation of Air Line Pilots' Associations: Tisdall, A.; **Gessl, M.; International Organization: Rahim, I.; International Maritime Organization for Standardization: Malesys, P.; International Source Supplies and Producers Association: Miller, J.J.; **Roughan, K.; United Nations Economic Commission for Europe: Kervella, O.; Universal Postal Union: Bowers, D.G.; World Nuclear Association: Gorlin, S.; World Nuclear Transport Institute: Green, L.

Waste Safety Standards Committee

Algeria: Abdenacer, G.; Argentina: Biaggio, A.; Australia: Williams, G.; *Austria:
Fischer, H.; Belgium: Blommaert, W.; Brazil: Tostes, M.; *Bulgaria:
Simeonov, G.; Canada: Howard, D.; China: Zhimin Qu; Croatia: Trifunovic, D.;
Cuba: Fernandez, A.; Cyprus: Demetriades, P.; Czech Republic: Lietava, P.;
Denmark: Nielsen, C.; Egypt: Mohamed, Y.; Estonia: Lust, M.; Finland: Hutri, K.;
France: Rieu, J.; Germany: Götz, C.; Ghana: Faanu, A.; Greece: Tzika, F.;
Hungary: Czoch, I.; India: Rana, D.; Indonesia: Wisnubroto, D.; Iran, Islamic

Republic of: Assadi, M.; *Zarghami, R.; Iraq: Abbas, H.; Israel: Dody, A.; Italy: Dionisi, M.; Japan: Matsuo, H.; Korea, Republic of: Won-Jae Park; *Latvia: Salmins, A.; Libyan Arab Jamahiriya: Elfawares, A.; Lithuania: Paulikas, V.; Malaysia: Sudin, M.; Mexico: Aguirre Gómez, J.; *Morocco: Barkouch, R.; Netherlands: van der Shaaf, M.; Pakistan: Mannan, A.; *Paraguay: Idoyaga Navarro, M.; Poland: Wlodarski, J.; Portugal: Flausino de Paiva, M.; Slovakia: Homola, J.; Slovenia: Mele, I.; South Africa: Pather, T. (Chairperson); Spain: Sanz Aludan, M.; Sweden: Frise, L.; Switzerland: Wanner, H.; *Thailand: Supaokit, P.; Tunisia: Bousselmi, M.; Turkey: Özdemir, T.; Ukraine: Makarovska, O.; United Kingdom: Chandler, S.; United States of America: Camper, L.; *Uruguay: Nader, A.; European Commission: Necheva, C.; European Nuclear Installations Safety Standards: Lorenz, B.; *European Nuclear Installations Safety Standards: Zaiss, W.; IAEA: Siraky, G. (Coordinator); International Organization for Standardization: Hutson, G.; International Source Suppliers and Producers Association: Fasten, W.; OECD Nuclear Energy Agency: Riotte, H.; World Nuclear Association: Saint-Pierre, S.



Where to order IAEA publications

In the following countries IAEA publications may be purchased from the sources listed below, or from major local booksellers. Payment may be made in local currency or with UNESCO coupons.

AUSTRALIA

DA Information Services, 648 Whitehorse Road, MITCHAM 3132 Telephone: +61 3 9210 7777 • Fax: +61 3 9210 7788 Email: service@dadirect.com.au • Web site: http://www.dadirect.com.au

BELGIUM

Jean de Lannoy, avenue du Roi 202, B-1190 Brussels Telephone: +32 2 538 43 08 • Fax: +32 2 538 08 41 Email: jean.de.lannoy@infoboard.be • Web site: http://www.jean-de-lannoy.be

CANADA

Bernan Associates, 4501 Forbes Blvd, Suite 200, Lanham, MD 20706-4346, USA Telephone: 1-800-865-3457 • Fax: 1-800-865-3450 Email: customercare@bernan.com • Web site: http://www.bernan.com

Renouf Publishing Company Ltd., 1-5369 Canotek Rd., Ottawa, Ontario, K1J 9J3 Telephone: +613 745 2665 • Fax: +613 745 7660 Email: order.dept@renoufbooks.com • Web site: http://www.renoufbooks.com

CHINA

IAEA Publications in Chinese: China Nuclear Energy Industry Corporation, Translation Section, P.O. Box 2103, Beijing

CZECH REPUBLIC

Suweco CZ, S.R.O., Klecakova 347, 180 21 Praha 9 Telephone: +420 26603 5364 • Fax: +420 28482 1646 Email: nakup@suweco.cz • Web site: http://www.suweco.cz

FINLAND

Akateeminen Kirjakauppa, PO BOX 128 (Keskuskatu 1), FIN-00101 Helsinki Telephone: +358 9 121 41 • Fax: +358 9 121 4450 Email: akatilaus@akateeminen.com • Web site: http://www.akateeminen.com

FRANCE

Form-Edit, 5, rue Janssen, P.O. Box 25, F-75921 Paris Cedex 19 Telephone: +33 1 42 01 49 49 • Fax: +33 1 42 01 90 90 Email: formedit@formedit.fr • Web site: http://www. formedit.fr

Lavoisier SAS, 145 rue de Provigny, 94236 Cachan Cedex Telephone: + 33 1 47 40 67 02 • Fax +33 1 47 40 67 02 Email: romuald.verrier@lavoisier.fr • Web site: http://www.lavoisier.fr

GERMANY

UNO-Verlag, Vertriebs- und Verlags GmbH, Am Hofgarten 10, D-53113 Bonn Telephone: + 49 228 94 90 20 • Fax: +49 228 94 90 20 or +49 228 94 90 222 Email: bestellung@uno-verlag.de • Web site: http://www.uno-verlag.de

HUNGARY

Librotrade Ltd., Book Import, P.O. Box 126, H-1656 Budapest Telephone: +36 1 257 7777 • Fax: +36 1 257 7472 • Email: books@librotrade.hu

INDIA

Allied Publishers Group, 1st Floor, Dubash House, 15, J. N. Heredia Marg, Ballard Estate, Mumbai 400 001, Telephone: +91 22 22617926/27 • Fax: +91 22 22617928 Email: alliedpl@vsnl.com • Web site: http://www.alliedpublishers.com

Bookwell, 2/72, Nirankari Colony, Delhi 110009 Telephone: +91 11 23268786, +91 11 23257264 • Fax: +91 11 23281315 Email: bookwell@vsnl.net

ITALY

Libreria Scientifica Dott. Lucio di Biasio "AEIOU", Via Coronelli 6, I-20146 Milan Telephone: +39 02 48 95 45 52 or 48 95 45 62 • Fax: +39 02 48 95 45 48 Email: info@libreriaaeiou.eu • Website: www.libreriaaeiou.eu

JAPAN

Maruzen Company, Ltd., 13-6 Nihonbashi, 3 chome, Chuo-ku, Tokyo 103-0027 Telephone: +81 3 3275 8582 • Fax: +81 3 3275 9072 Email: journal@maruzen.co.jp • Web site: http://www.maruzen.co.jp

REPUBLIC OF KOREA

KINS Inc., Information Business Dept. Samho Bldg. 2nd Floor, 275-1 Yang Jae-dong SeoCho-G, Seoul 137-130 Telephone: +02 589 1740 • Fax: +02 589 1746 • Web site: http://www.kins.re.kr

NETHERLANDS

De Lindeboom Internationale Publicaties B.V., M.A. de Ruyterstraat 20A, NL-7482 BZ Haaksbergen Telephone: +31 (0) 53 5740004 • Fax: +31 (0) 53 5729296 Email: books@delindeboom.com • Web site: http://www.delindeboom.com

Martinus Nijhoff International, Koraalrood 50, P.O. Box 1853, 2700 CZ Zoetermeer Telephone: +31 793 684 400 • Fax: +31 793 615 698 Email: info@nijhoff.nl • Web site: http://www.nijhoff.nl

Swets and Zeitlinger b.v., P.O. Box 830, 2160 SZ Lisse Telephone: +31 252 435 111 • Fax: +31 252 415 888 Email: infoho@swets.nl • Web site: http://www.swets.nl

NEW ZEALAND

DA Information Services, 648 Whitehorse Road, MITCHAM 3132, Australia Telephone: +61 3 9210 7777 • Fax: +61 3 9210 7788 Email: service@dadirect.com.au • Web site: http://www.dadirect.com.au

SLOVENIA

Cankarjeva Zalozba d.d., Kopitarjeva 2, SI-1512 Ljubljana Telephone: +386 1 432 31 44 • Fax: +386 1 230 14 35 Email: import.books@cankarjeva-z.si • Web site: http://www.cankarjeva-z.si/uvoz

SPAIN

Díaz de Santos, S.A., c/ Juan Bravo, 3A, E-28006 Madrid Telephone: +34 91 781 94 80 • Fax: +34 91 575 55 63 Email: compras@diazdesantos.es, carmela@diazdesantos.es, barcelona@diazdesantos.es, julio@diazdesantos.es Web site: http://www.diazdesantos.es

UNITED KINGDOM

The Stationery Office Ltd, International Sales Agency, PO Box 29, Norwich, NR3 1 GN Telephone (orders): +44 870 600 5552 • (enquiries): +44 207 873 8372 • Fax: +44 207 873 8203 Email (orders): book.orders@tso.co.uk • (enquiries): book.enquiries@tso.co.uk • Web site: http://www.tso.co.uk

On-line orders

DELTA Int. Book Wholesalers Ltd., 39 Alexandra Road, Addlestone, Surrey, KT15 2PQ Email: info@profbooks.com • Web site: http://www.profbooks.com

Books on the Environment Earthprint Ltd., P.O. Box 119, Stevenage SG1 4TP Telephone: +44 1438748111 • Fax: +44 1438748844 Email: orders@earthprint.com • Web site: http://www.earthprint.com

UNITED NATIONS

Dept. 1004, Room DC2-0853, First Avenue at 46th Street, New York, N.Y. 10017, USA (UN) Telephone: +800 253-9646 or +212 963-8302 • Fax: +212 963-3489 Email: publications@un.org • Web site: http://www.un.org

UNITED STATES OF AMERICA

Bernan Associates, 4501 Forbes Blvd., Suite 200, Lanham, MD 20706-4346 Telephone: 1-800-865-3457 • Fax: 1-800-865-3450 Email: customercare@bernan.com · Web site: http://www.bernan.com

Renouf Publishing Company Ltd., 812 Proctor Ave., Ogdensburg, NY, 13669 Telephone: +888 551 7470 (toll-free) • Fax: +888 568 8546 (toll-free) Email: order.dept@renoufbooks.com • Web site: http://www.renoufbooks.com

Orders and requests for information may also be addressed directly to:

Marketing and Sales Unit, International Atomic Energy Agency

Vienna International Centre, PO Box 100, 1400 Vienna, Austria Telephone: +43 1 2600 22529 (or 22530) • Fax: +43 1 2600 29302 Email: sales.publications@iaea.org • Web site: http://www.iaea.org/books

RELATED PUBLICATIONS



FUNDAMENTAL SAFETY PRINCIPLES IAEA Safety Standards Series No. SF-1 STI/PUB/1273 (21 pp., 2006) ISBN 92-0-110706-4	Price: €25.00
SAFETY OF RESEARCH REACTORS IAEA Safety Standards Series No. NS-R-4 STI/PUB/1220 (121 pp., 2005) ISBN 92-0-115804-1	Price: €24.00
COMMISSIONING OF RESEARCH REACTORS IAEA Safety Standards Series No. NS-G-4.1 STI/PUB/1268 (65 pp.; 2006) ISBN 92-0-109606-2	Price: €28.00
MAINTENANCE, PERIODIC TESTING AND INSPECTION OF RESEARCH REACTORS IAEA Safety Standards Series No. NS-G-4.2 STI/PUB/1270 (67 pp., 2006) ISBN 92-0-109806-5	Price: €30.00
CORE MANAGEMENT AND FUEL HANDLING FOR RESEARCH REACTORS IAEA Safety Standards Series No. NS-G-4.3 STI/PUB/1315 (53 pp.; 2008) ISBN 978–92–0–111307–8	Price: €22.00
OPERATIONAL LIMITS AND CONDITIONS AND OPERATING PROCEDURES FOR RESEARCH REACTORS IAEA Safety Standards Series No. NS-G-4.4 STI/PUB/1333 (67 pp.; 2008) ISBN 978–92–0–103508–0	Price: €23.00
THE OPERATING ORGANIZATION AND THE RECRUITMENT, T AND QUALIFICATION OF PERSONNEL FOR RESEARCH REAC IAEA Safety Standards Series No. NS-G-4.5 STI/PUB/1335 (89 pp.; 2008) ISBN 978–92–0–103708–4	RAINING CTORS Price: €20.00
RADIATION PROTECTION AND RADIOACTIVE WASTE MANAGE IN THE DESIGN AND OPERATION OF RESEARCH REACTORS IAEA Safety Standards Series No. NS-G-4.6 STI/PUB/1369 (104 pp.; 2009) ISBN 978-92-0-111608-6	BEMENT
AGEING MANAGEMENT FOR NUCLEAR POWER PLANTS IAEA Safety Standards Series No. NS-G-2.12 STI/PUB/1373 (48 pp.; 2009) ISBN 978–92–0–112408–1	Price: €20.00

Safety through international standards

The fundamental safety objective is to protect people and the environment from harmful effects of ionizing radiation.

This fundamental safety objective of protecting people — individually and collectively — and the environment has to be achieved without unduly limiting the operation of facilities or the conduct of activities that give rise to radiation risks.

> Fundamental Safety Principles: Safety Fundamentals, IAEA Safety Standards Series No. SF-1 (2006)

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA ISBN 978–92–0–102810–5 ISSN 1020–525X